DECLARATION OF COVENANTS, CONDITIONS & RESTRICTIONS FOR HAGAR COURT CONDOMINIUMS

CITY OF SANTA CRUZ

COUNTY OF SANTA CRUZ, STATE OF CALIFORNIA

THIS DECLARATION is made	, 2003, by THE REGENTS
OF THE UNIVERSITY OF CALIFORNIA,	a California public corporation ("Declarant")

Declarant is the owner of that certain real property located in the City of Santa Cruz, County of Santa Cruz described in Exhibit A to this Declaration (the "Land"). Declarant has improved the Land with fifty (50) dwelling units and one (1) administrative unit and other facilities. By this Declaration, Declarant intends to establish a plan of condominium ownership for the Project, pursuant to which an undivided fractional leasehold interest underlying each owner's fee interest in a Condominium will be transferred to each such owner.

Consistent with the requirements of section 1351(e) of the California Civil Code, Declarant has consented to the recordation of a condominium plan for the Project and has caused such Condominium Plan to be duly executed on behalf of Declarant.

DECLARATION

Declarant declares that the Project is, and shall be, held, conveyed, hypothecated, encumbered, leased, rented, used and occupied subject to the following limitations, reservations, restrictions, easements, covenants, conditions, servitudes, liens and charges, all of which are declared and agreed to be in furtherance of a plan of condominium ownership as described in California Civil Code, sections 1350 through 1372, for the subdivision, improvement, protection, maintenance, and sale of Condominiums within the Project, and all of which are declared and agreed to be for the purpose of enhancing, maintaining and protecting the value and attractiveness of the Project. All of the limitations, restrictions, easements, reservations, covenants, conditions, servitudes, liens and charges shall run with the land, shall be binding on and inure to the benefit of all parties having or acquiring any right, title or interest in the Project, and shall be binding on and inure to the benefit of the successors in interest

of such parties. Declarant further declares that it is the express intent that this Declaration satisfy the requirements of California Civil Code section 1355.

ARTICLE 1

DEFINITIONS, EXHIBITS AND INTERPRETATION

1.1 Definitions. As used in this Declaration:

"Act" means sections 1350 through 1372 of the California Civil Code, as now in effect and as the same may from time to time be amended, replaced or supplemented.

"Architectural Review Board" or "ARB" means the architectural review board established pursuant to Section 16.2 hereof.

"Articles" means the Association's Articles of Incorporation, as the same may from time to time be amended.

"Association" means the Hagar Court Condominiums Association, a California nonprofit mutual benefit corporation, and its successors and assigns.

"Association Rules" means the rules and regulations, adopted by the Board from time to time regulating the use and enjoyment of the Common Areas.

"Board of Directors" or "Board" means the Board of Directors of the Association.

"Bylaws" means the Association's Bylaws, as the same may from time to time be amended.

"Common Areas" means the leasehold interest in the entire Project created by the Common Areas Lease, excepting therefrom all Units as defined in this Declaration, as shown on the Condominium Plan. The Common Areas shall include, without limitation, the Project's structures (except for the Units), the leasehold rights in the Land upon which the structures are located, the air space above such structures, all bearing walls, columns, floors, roofs, slabs, foundations, common stairways and hallways, reservoirs, tanks, pumps, air ducts, and other central services equipment, pipes, ducts, flues, chutes, conduits, wires, and other utility installations, wherever located, servicing more than one Unit (except the outlets thereof when located within and servicing only a particular Unit), lawns, pavement, trees, all other landscaping on the Land on which the

Project's structures are located, parking and storage spaces, if any, and the structures, railings and enclosures of balconies, decks, private yards and entryways.

"Common Areas Lease" means that certain	Common Areas Lease
dated, by and between The Regents o	f the University of California and the
Association as lessee subject to various conditions, including the Association's consent	
to this Declaration, a copy of which Common Areas Lease was recorded on	
, 2003, as Instrument No	, Official Records, Santa Cruz
County Recorder, Santa Cruz County, California.	

"Condominium" means an estate in real property, as defined in sections 783 and 1351(f) of the California Civil Code, consisting of (i) a Unit; (ii) a Percentage Interest in the Common Areas; (iii) any exclusive or nonexclusive right of entry, right of possession, easement or easements appurtenant to such Unit over the Common Areas or other areas as described in this Declaration, the Condominium Plan and the deed to the Condominium; and (iv) a membership in the Association. Whenever reference to a Condominium is made in this Declaration, the Condominium Plan, any deed, or elsewhere, it shall be assumed that such reference is made to the Condominium as a whole, including each of its component elements.

"Condominium Instruments" means this Declaration, the Articles, and the Bylaws, together with any supplements, amendments, exhibits, schedules, or certifications thereto.

"Condominium Plan" means the Amended Map for the Condominium Subdivision for the Project filed on October 17, 2003, Volume 104 of Maps, Page 31 in the official records of the County Recorder of Santa Cruz, California, pursuant to section 1351(e) of the California Civil Code. A copy of the Condominium Subdivision is attached as Exhibit B. In the event of any conflict between Exhibit B and the recorded Condominium Subdivision, the recorded document shall control.

"Convey", when applied to a Condominium, means the due execution of a valid deed to a Unit by an owner and an assignment of such owner's leasehold interest in the Common Areas.

"Current Dollars" means any dollar amount stated herein multiplied by the quotient obtained by dividing (i) the level of the "National Consumer Price Index for All Urban Consumers, All Items" (1982-84=100) (commonly known as the "CPI-U") prepared and published by the United States Department of Labor, Bureau of Labor Statistics, as of the date of computation by (ii), the level of the CPI-U as of the date of

recordation of this Declaration; provided, however, that should the CPI-U cease to be published, the above-described computation shall be made using such other successor or reasonably comparable index as shall be determined by (A) Declarant during the Declarant Control Period, and (B) the Ground Lessor following termination of the Declarant Control Period.

"Declarant" means The Regents of the University of California, a California corporation, and its successors and assigns, if such successors and assigns are duly assigned the rights of Declarant pursuant to Section 19.2 of this Declaration.

"Declarant Control Period" means the period of time extending from the recordation of this Declaration to the earliest of (i) that date upon which more than seventy-five percent (75%) of the Condominiums which may be included in the Project have been conveyed to owners other than Declarant; (ii) the date two (2) years after the first Condominium has been conveyed to an owner other than Declarant; or (iii) the date upon which Declarant notifies the owners that Declarant relinquishes its rights under this Declaration and the Bylaws to appoint and remove a majority of the members of the Board of Directors or to exercise powers and responsibilities otherwise assigned by the Condominium Instruments to the Association, the Board of Directors, or any officer of the Association.

"Declarant Unit" means Unit 51 and the associated Condominium, the ownership of which Declarant intends to retain following expiration of the Declarant Control Period.

"Declaration" means this Declaration of Covenants, Conditions, and Restrictions and includes, as the context requires, its amendments, modifications, or supplements.

"Domestic Partner" means a person who (i) is a member of a domestic partnership registered with the State of California, or (ii) meets all of the criteria for domestic partnership as described in the University of California Retirement Plan.

"Exclusive Use Common Areas" means those portions of the Common Areas which are for the exclusive use of the owners of particular Units in accordance with the California Civil Code, section 1351(i), as described in Article 3 of this Declaration. Such Exclusive Use Common Areas include parking stalls and storage spaces. Each such Exclusive Use Common Area is identified on the Condominium Plan by the Identifying Number of the Condominium whose owner enjoys an easement for the use of such Exclusive Use Common Areas, followed by the letter designation, "P", "P-101," or "P-

102" for parking stalls "S" for storage spaces, "WH/F" for water heater/furnace sheds, and "Y" for yards.

"Ground Lessor" means The Regents of the University of California, its successors or assigns, in its capacity as Lessor under the Common Areas Lease.

"Identifying Number" means one or more letters or numbers included in the Condominium Plan that identify only one Condominium in the Project.

"Land" means the real property, exclusive of any improvements thereon, as described in Exhibit A to this Declaration, leased for the purpose of creating leasehold interests to be submitted to this Declaration.

"Member" means every person or entity holding a membership in the Association.

"Mortgage" means a mortgage or deed of trust encumbering a Condominium or other portion of the Project. A "First Mortgage" means a Mortgage having priority of security over all other Mortgages encumbering the same Condominium.

"Mortgagee" includes the beneficiary under a deed of trust. An "Institutional Mortgagee" means a Mortgagee that is (i) a bank or savings and loan association or mortgage company, or other entity chartered or licensed under federal or state laws whose principal business is lending money on the security of real property or investing in such loans; (ii) any insurance company; (iii) any private or governmental agency or instrumentality, including without limitation, the Federal National Mortgage Association and the Federal Home Loan Mortgage Corporation, engaged in secondary market transactions with respect to real estate loans; and (iv) Declarant. A "First Mortgage," "First Mortgagee," or "First Institutional Mortgagee" means the beneficiary of a First Mortgage.

"Owner" means each person or entity holding a record ownership interest in a Condominium, including Declarant. "Owner" shall not include any person or entity who holds an interest in a Condominium merely as security for the performance of an obligation.

"Percentage Interest" means the percentage of undivided leasehold interest in all or a portion of the Common Areas pursuant to the provisions of Section 2.3 of this Declaration.

"Person" means a natural person, corporation, partnership, association, trust, or other entity capable of holding title to real property, or any combination thereof.

"Project" means the Land and all improvements on the Land, as subjected to the Common Areas Lease, collectively known as Hagar Court Homes, reserving therefrom Declarant's reversionary interest in the Land and improvements, as further described in Section 2.1 of this Declaration.

"Rules and Regulations" means those rules and regulations adopted by the Association or its Board, including any amendments thereto governing the use and occupancy of the Project.

"Single Family" means one or more persons, each related to the other by blood, marriage or adoption, or a group of not more than six (6) persons, not so related, together with his, hers, or their domestic servants, maintaining a common household.

"Unit" means a separate estate in space as defined in California Civil Code section 1351(f) and as described in Section 3.1 of this Declaration. A Unit consists of all those component elements bearing the same Identifying Number designation in the Condominium Plan. The Identifying Number designation of a component element coincides with the number of that Unit of which it is a part. Whenever reference is made to any of the Units, it shall be construed that reference is made to the Unit as a whole and to each and all of its component elements. The Units and their respective boundaries are further described in Section 3.1 of this Declaration, in the Condominium Plan, and in deeds conveying Condominiums, except that in interpreting deeds and plans, the existing physical boundaries of a Unit or of a Unit reconstructed in substantial accordance with the original plan shall be conclusively presumed to be its boundaries, rather than the description expressed in the deed or Condominium Plan, regardless of minor variance between boundaries shown on the Condominium Plan, or in the deed and those of the building and regardless of settling or lateral movement of the building.

- **1.2** Exhibits. All exhibits to this Declaration and all attachments thereto are hereby incorporated by reference in this Declaration.
- 1.3 <u>Interpretation of Declaration.</u> The terms, covenants, conditions, and restrictions set forth in this Declaration, the Bylaws, and the Articles are intended to be construed to be consistent with the terms, covenants, conditions, and restrictions set forth in the Common Areas Lease. In the event of any conflict in interpretation, the

terms, covenants, conditions and restrictions contained in the Common Areas Lease shall control.

ARTICLE 2

ESTABLISHMENT OF THE CONDOMINIUM

- 2.1 Submission of Property and Division into Estates.
- (a) Declarant does hereby submit and subject the leasehold estate in the Common Areas created by the Common Areas Lease, together with all improvements, easements, and rights created pursuant thereto, to the provisions of the Act and to this Declaration, and hereby creates with respect to such property Condominiums to be known as "Hagar Court Homes."
- (b) An Owner shall not be entitled to sever his or her Condominium from membership in the Association and shall not be entitled to sever his or her Condominium and membership from his or her undivided interest in all or a portion of the Common Areas for any purpose. None of the component interests in a Condominium can be severally sold, conveyed, encumbered, hypothecated, or otherwise dealt with, and no violation or attempted violation of this provision shall be effective. No Owner can sever any easement appurtenant to his or her Unit over the Common Areas from his or her Condominium, and any attempt to do so shall be void. It is intended by this provision to restrict severability under section 1358 of the California Civil Code, as now in effect or as hereinafter amended or replaced.
- (c) Any conveyance of a Unit or any portion of it by an Owner shall be presumed to convey the entire Condominium of which such Unit is an element.
- (d) Every contract of sale, deed, lease, mortgage, deed of trust, or other instrument recorded in the Official Records of the County Recorder of Santa Cruz County, California, and affecting any Condominium created hereby shall describe it by its Identifying Number, followed by the name "Hagar Court Homes," followed by specific reference to the Condominium Plan. Every such contract or sale, deed, lease, mortgage, deed of trust, or other instrument shall expressly indicate that it is subject to and subordinate to this Declaration. As used in this subsection (d), "specific reference" means book and page recording data from the Official Records of the County Recorder of Santa Cruz County, California.

- 2.2 Recordation of Condominium Plan. Prior to conveyance of any Condominium, Declarant shall file and record the Condominium Plan in the Office of the County Recorder of Santa Cruz County, California, pursuant to the requirements of section 1351(e) of the California Civil Code. Such Condominium Plan shall contain all of the information required by and otherwise conform to the requirements of said section 1351(e).
- 2.3 Assignment of Percentage Interest. Each Owner of a Condominium in the Project shall have a percentage of undivided leasehold interest in the Common Areas that shall be determined by dividing the total square footage of floor space in such Unit by the total square footage of floor space of all Units, such result to be carried to the fourth decimal place, as set forth in Exhibit C hereto. Each such Owner's percentage of undivided leasehold interest in the Common Areas may not be altered or changed, and shall be specified in the deed from Declarant to each such Owner.

ARTICLE 3

UNITS, EASEMENTS AND RESERVATIONS

3.1 Units.

- (a) The boundaries of each Unit (including any sub-elements thereof as may be designated on the Condominium Plan) shall be the interior surfaces of the perimeter walls, floors, ceilings, windows and doors, where they exist and otherwise to the inclined, vertical or horizontal planes, or curved surfaces, as the case may be, at the limits of the dimensions shown on the Condominium Plan. The Unit includes both the portions of the building so described, and not excluded in Section 3.1(c) hereof, and the air space so encompassed. The Unit includes all fixtures, outlets, appliances, heaters, and similar devices located within and servicing only the particular Unit, or such device, wherever located, if designated on the Condominium Plan as a part of the Unit.
- (b) The boundaries of that portion of a Unit designated on the Condominium Plan as a deck, private yard, patio or entryway, shall be the adjoining exterior surfaces of the walls, windows, doors, overhangs and ceilings of the structure of the Common Area and where the deck, patio or entryway is not so adjoined, the boundaries are the inclined, vertical or horizontal planes, or curved surfaces, as the case may be, at the limits of the dimensions for the particular deck, patio or entryway shown on the Condominium Plan. The deck, patio or entryway shall include only the air space and finished floor or ground surface within, but not the other portions of the Project, Common Areas and surfaces thereof described by or contained within the boundaries of the deck, patio or entryway.

(c) The following are not part of a Unit: bearing walls, columns, sub-floors, roofs, slabs, foundations, common stairways and hallways, reservoirs, tanks, pumps, air ducts and other central services equipment, pipes, ducts, flues, chutes, conduits, wires and other utility installations, wherever located, servicing more than one Unit (except the outlets thereof when located within and servicing only a particular Unit), parking spaces and storage spaces, and the railings and enclosures of decks, patios or entryways.

3.2 Nonexclusive Easements.

- (a) Every Owner has a nonexclusive easement of use, enjoyment, ingress, egress, and support in, to, and throughout the Common Areas and any improvements or facilities on those Common Areas. Such nonexclusive easements, however, shall be subordinate to, and shall not interfere with, exclusive easements over the Exclusive Use Common Areas. Each such nonexclusive easement shall be appurtenant to the respective Unit and shall pass, as part of the Condominium, with the title to the Unit. Nonexclusive easements also shall be subject to the following rights and restrictions:
- (i) The right of the Association to adopt and to enforce the Association Rules and Regulations, including but not limited to, rules regulating the number of guests in and the use of the Common Areas, subject to Sections 4.1 and 7.5 of this Declaration.
- (ii) Subject to the provisions of this Declaration, the right of the Association to borrow money to improve, repair, or maintain the Common Areas.
- (iii) The right of the Association to assign, rent, license, or otherwise designate and control use of any unassigned parking and storage spaces within, and any recreational facility situated upon, the Common Areas, and to charge reasonable fees for admission and use.
- (iv) The right of the Association to suspend the right of an Owner to use any recreational or other facility in the Common Areas as provided in Section 7.6 of this Declaration.
- (v) The rights of Declarant, until termination of the Declarant Control Period, and, thereafter, the Association, to assign or delegate to Declarant or any governmental entity, including Declarant, or to contract with any private security patrol

company to provide police or security service in the Project, or to maintain any driveways or paved areas within the Common Areas.

- (vi) The right of Declarant, until termination of the Declarant Control Period, and, thereafter, the Association, to grant licenses, permits and easements pursuant to Section 3.5, below.
- (b) Each Condominium (including specifically the Unit) or the Common Areas, as the case may be, shall be subject to the following rights of entry and use:
- (i) The right of Declarant or its designees to enter upon any portion of the Project to construct the improvements thereon and to make repairs, provided that such entry shall not unreasonably interfere with the use or occupancy of any occupied Unit unless authorized by its Owner, which authorization shall not be unreasonably withheld. This right of entry shall be immediate in the case of an emergency originating upon or threatening any Unit, whether or not its Owner is present.
- (ii) The right of Declarant to modify, alter, remove, or improve defective, obsolete, or nonfunctional portions of the Common Areas, including without limitation, any equipment, fixtures and appurtenants, when in Declarant's sole judgment it is necessary or desirable to do so, until the later of the expiration of any applicable warranty period or the Declarant Control Period.
- (iii) The right of the Association, or its agents, to enter any Unit to cure any violation or breach of this Declaration, the Bylaws, or the Association Rules and Regulations, provided that (except in cases of emergency) at least fifteen (15) calendar days prior to such entry, written notice of such violation or breach has been given to the Owner and such Owner has not cured such violation or breach within the fifteen (15) day period after notice is given. The Association shall be entitled to levy a special assessment for its costs of effecting such cure against the Owner in accordance with the procedures set forth in Section 9.7 of this Declaration. These rights of entry and cure shall be immediate in case of an emergency originating upon or threatening any Unit, whether or not its Owner is present.
- (iv) The right of the Association, or its agents, to enter any of the Units or the Common Areas to perform its obligations and duties under this Declaration, including obligations or duties with respect to construction, maintenance, or repair of facilities located in the Common Areas, or that affect more than one Unit; watering, planting, cutting, removing, and otherwise caring for the landscaping upon the Common Areas; and cleaning, repairing, replacing, and otherwise maintaining or

causing to be maintained utility lines serving each Unit. Such entry shall not unreasonably interfere with the use or occupancy of any occupied Unit unless authorized by its Owner, which authorization shall not be unreasonably withheld. This right of entry shall be immediate in the case of an emergency originating upon or threatening any Unit, whether or not its Owner is present.

3.3 Exclusive Use Common Areas.

- (a) Each Owner whose Condominium's Identifying Number appears on a parking space or storage area designated on the Condominium Plan shall have an exclusive easement for vehicular parking purposes designated as a parking space and for storage of personal property designated as a storage area over that portion of the Common Areas designated on such plan. Each Owner having such an exclusive easement shall, at his or her own expense, keep his or her space or spaces in a neat, clean, attractive, and safe condition at all times.
- (b) Each Owner shall have an exclusive easement to use and enjoy that portion of the Common Areas, , adjacent to his or her Unit designated on the Condominium Plan as a "Yard," "Storage," or "Water Heater/Furnace", and bearing his or her Condominium's Identifying Number. Each such Owner shall, at his or her own expense, keep each such patio or balcony in a neat, clean, attractive, and safe condition at all times. The Association shall maintain, repair and replace any wall or fence which surrounds or adjoins any Private Yard and shall assess each Owner of a Unit which is appurtenant to such Private Yard for the cost of such maintenance, repair or replacement, as an addition to the regular assessment of such Unit. Two or more Owners may share responsibility, in the proportion determined by the Board, for such cost.
- 3.4 Encroachments. If any portion of a Unit encroaches on the Common Areas, except if such encroachment is caused by the willful act of the Owner of such Unit, or if any portion of the Common Areas encroaches on any Unit, a valid easement exists for each such encroachment and for the maintenance of it as long as it remains, and all Units and the Common Areas are made subject to such easements. If any structure containing a Unit is partially or totally destroyed and then rebuilt and any encroachment on the Common Areas results, a valid easement exists for such encroachment and for the maintenance of it as long as it remains, and all Units and the Common Areas are made subject to such easements.
- **3.5** <u>Power to Grant Easements.</u> Until termination of the Declarant Control Period, Declarant, and thereafter the Association, shall have the power to grant and

convey in the name of all the Owners as their attorney-in-fact (or in the name of the Association as to any property to which the Association holds title) to any Owner or other party easements, licenses, permits, and rights-of-way in, on, over, or under the Common Areas as follows:

(i) In the case of an Owner, for the purpose of establishing Exclusive Use Common Areas in favor of an Owner.

(ii) In the case of other parties:

- (A) For the purpose of constructing, erecting, operating, or maintaining lines, cables, wires, conduits, or other devices for electricity, cable television, power, telephone, and other services.
- (B) For public sewers, storm water drains and pipes, water systems, sprinkling systems, water, heating and gas lines or pipes, and any similar public or quasi-public improvements or facilities.

(C) For emergency access.

Each Owner, by accepting a deed to a Condominium, expressly consents to such easements, licenses, permits, and rights-of-way and authorizes and appoints Declarant, until termination of the Declarant Control Period, and thereafter the Association, as attorney-in-fact of such Owner to execute any and all instruments conveying or creating such easements, licenses, permits, or rights-of-way, provided that no such easement may be granted if it would permanently interfere with the occupancy by any Owner of his or her Unit. Each Owner is subject to all rights and duties assigned to Owners hereunder, except as specifically provided to the contrary in the Common Areas Lease with respect to Declarant.

3.6 Reservations for Sales and Management Purposes.

- (a) Declarant reserves the right to use any Unit owned or leased by Declarant as a model, management office, sales office, or customer service office for the Project. Declarant may also, from time to time, relocate such model, management office, sales office, or customer service office within the Project and, upon such relocation, may remove the furnishings thereof.
- (b) Declarant reserves the right to maintain on the Common Areas no more than six (6) signs for the purposes set forth in Section 3.6(a) of this Declaration, each

having a maximum face area of four (4) square feet, which signs may be placed at any location on the Common Areas and may be relocated or removed at the sole discretion of Declarant, provided that any such sign is not located so as to restrict unreasonably the access of Owners to the Common Areas or their Units.

- (c) The reservations set forth in this Section 3.6 shall remain in effect until the time that Declarant has conveyed all Condominiums to Unit Owners other than Declarant.
- 3.7 <u>Reservation for Rental Purposes.</u> Declarant reserves the right, for itself and its successors and assigns, to own one or more Condominiums for rental purposes and to lease such Condominiums to tenants acceptable to Declarant.

ARTICLE 4

USE RESTRICTIONS

In addition to all other covenants, restrictions and limitations contained herein, the use of the Project, each Unit therein, and the Common Areas is subject to the following restrictions:

4.1 Business Usage Prohibited. Except as expressly permitted herein, no Unit other than the Declarant Unit, shall be used for other than Single Family residential purposes. Residential purposes could include family child day care, or other children's activity facility located within the Project, approved by the Association, and serving children of Hagar Court residents. No part of the Project or any Unit, other than the Declarant Unit, shall ever be used or allowed, authorized or caused to be used in any way, directly or indirectly, for business, commercial, manufacturing, mercantile, storage, or other nonresidential purpose, except for business usage which would be permitted under the Ordinances of the City of Santa Cruz with regard to home occupations, or for professional or academic endeavors not requiring the continuing presence of any employee or business invitee. However, gainful employment consisting of individual use by an Owner of telecommunications devices shall not violate the restriction in this Section. Declarant, its successors or assigns, may use the Units owned by it as models, management offices, sales offices, or customer service offices during the construction and sales period of the Project and thereafter may use the Declarant Unit for such business of UCSC as Declarant, in its sole, but reasonable discretion, shall determine is appropriate and compatible with the residential character of Hagar Court Homes.

4.2 <u>Liquid Furniture</u>. No waterbed or other item of liquid-filled furniture shall be placed in any portion of any Unit unless such item shall conform to the requirements if any issued by the ARB with respect to such furniture and unless, at the option of the Association, the Owner shall provide evidence of insurance as described in section 1940.5 of the California Civil Code.

4.3 [NOT USED]

- 4.4 <u>Signs and Flags.</u> Except for signs approved by the ARB for the benefit of the Project, political campaign signs displayed within forty-five (45) calendar days prior to an election, or temporary signs approved in advance by the Board, no signs of any kind shall be displayed in public view on or about the exterior of any Unit, except signs not larger than is reasonable and customary in the area advertising such Unit for sale or lease. Display of a flag shall be subject to the Association's Rules and Regulations. The Association shall have the right to remove an item displayed in violation of this Section 4.4.
- **4.5** <u>Children.</u> Each Owner shall be accountable to the remaining Owners, their families, servants, guests, tenants and invitees for the conduct and behavior of his or her children and other children residing in or visiting his or her Unit.
- 4.6 Pets. Each Owner shall comply with the "Campus Pet Policy" as amended from time to time. Pets shall not be allowed in the Common Areas except as permitted by the Rules and Regulations adopted by the Board. Each Owner shall be absolutely liable to each and all remaining Owners, their families, servants, guests, tenants and invitees for any damage to person or property caused by any pet brought into or kept upon or in the Project by an Owner or his or her family, servants, guests, tenants and invitees. Except as provided in this Section 4.6, no animals of any kind shall be brought within the Project or kept in or on any Unit. The Association may prohibit the keeping of any animal that in the sole and exclusive opinion of the Board constitutes a nuisance to any other Owner. Owner to comply with Campus Pet Policy.
- 4.7 Offensive Activities. No Owner shall permit or suffer anything to be done or kept upon or in his or her Unit or in the Common Areas which will increase the rate of insurance thereon or will obstruct or interfere with the rights of other Owners, their families, servants, guests, tenants and invitees, nor annoy them by unreasonable noises, vibrations, bright or flashing lights or otherwise, nor shall in any way interfere with the quiet enjoyment of each of the Owners of his or her respective Unit, nor will he or she commit or permit any nuisance, noxious, illegal or offensive activity to be permitted thereon or therein.

- 4.8 Owner Negligence. Each Owner shall be liable to the Association for any damage to the Common Areas or any equipment thereon which may be sustained by reason of the negligence or intentional acts or omissions of any said Owner, his or her family, servants, guests, tenants and invitees, to the extent that any such damage shall not be covered by insurance.
- **4.9** <u>Trash Disposal.</u> Trash, garbage, recyclable materials, or other waste shall be kept only in sanitary containers. No Owner shall permit or cause any trash or refuse to be kept within any portion of the Project other than in customary receptacles. These receptacles shall be located only in places specifically designated for such purpose.
- **4.10** Outside Laundering and Drying. No exterior clothesline shall be erected or maintained, and there shall be no exterior drying or laundering of clothes on balconies, patios, porches, or other outside areas other than on drying "trees" or other removable apparatus specifically approved by the ARB.
- **4.11** Prohibited Restrictions. No Owner shall execute or file for record any instrument which imposes restrictions upon the sale, leasing or occupancy of his or her Unit other than as set forth in this Declaration, including, but not limited to restrictions based on race, color, religion, sex, sexual orientation, age, marital status, or national origin.
- 4.12 <u>Storage.</u> No Owner shall park or store machinery, equipment, baby carriages, playpens, bicycles, wagons, benches or chairs on any part of the Common Areas, except that such personal property may be stored in storage areas, if any, that may be designated by the Association for that purpose. Sun decks and other recreational areas may be used for their customary purposes.
- 4.13 Antenna and Clothesline. No television or radio poles, antenna, satellite transmission or reception equipment, flagpoles, exterior alarms, or external fixtures other than those originally installed by Declarant or approved by the ARB, and any replacements thereof, shall be constructed, erected or maintained on or within the Project or any structures on it. No wiring, insulation, air conditioning or other machinery or equipment other than that originally installed by Declarant or approved by the ARB, and any replacements thereto shall be constructed, erected or maintained on or within the property or any structures on it. Each Owner shall have the right to maintain television or radio antenna within completely enclosed portions of his or her Unit. The location of common antenna or connection facilities for cable television shall be solely as designated by the ARB.

- 4.14 Personal Business Records Permitted. No restriction contained in this Article shall be construed in such a manner as to prohibit any Owner from the use of a Unit to (a) maintain his or her personal, professional library; (b) keep his or her personal business records or accounts; or (cconduct his or her personal or professional telecommunication activities. Such uses are expressly declared to be customarily incident to the residential use of the Project and not in violation of any provision of this Article.
- 4.15 <u>Vehicles and Parking</u>. Unless otherwise specifically permitted by the Association, no motor vehicle shall be parked or left within the Project other than in an assigned or appurtenant parking space or in any designated guest parking space or area. No boat, trailer, recreation vehicle, camper, or commercial vehicle shall be parked or left within the Project other than in a parking area designated by the Association for the parking or storage of such vehicles, and no such vehicles may be used for habitation or recreation while within the Project. No truck or recreation vehicle may be stored within the Project unless it is the Owner's principal means of transportation. No vehicular repairs shall be undertaken within the Project.
- 4.16 Restrictions Applicable to Guests, etc. The use of any Unit or of the Common Areas or any portion thereof by any guest, invitee or tenant of any Owner shall be subject to all of the provisions of this Declaration, including, without limitation, all of the use restrictions imposed under this Article, the Bylaws and the Rules and Regulations of the Association, and the Association may proceed directly against such guest, invitee or tenant in the enforcement of the provisions of this Declaration, the Association's Bylaws or the Association's Rules and Regulations.
- 4.17 <u>Fences and Screens.</u> Except as may from time to time be provided in the Property Use and Maintenance Regulations, no fences, awnings, ornamental screens, screen doors, sunshades or walls of any nature shall be erected or maintained on or around any portion of any structure or elsewhere within the Project, except those that are installed in accordance with the original construction of the Project, and their replacements or as are authorized and approved by the ARB.
- **4.18** Gas or Liquid Storage. No tank for the storage of gas or flammable liquid shall be installed upon or in the Project unless such installation is done by Declarant, the Association, or has been approved by the Board.
 - **4.19** <u>Potted Plants.</u> No potted plants shall be placed on stucco surfaces.
- **4.20** <u>Diseased Plants.</u> No plants or seeds infected with noxious insects or plant diseases shall be brought upon, grown or maintained within the Project.

- **4.21** <u>Climbing Vines.</u> No Owner shall plant vines or creepers to climb or entwine any portion of any structure or elsewhere within the Project, including, but not limited to, deck railings and fencing.
- 4.22 <u>Plants in Common Areas</u>. No Owner shall plant, cut, trim, prune, remove, replace or otherwise alter or affect the appearance or location of any tree, plant or other vegetation located in any portion of the Common Areas without the prior written consent of the ARB. The Association may recover from any Owner violating this Section 4.22 the cost of restoring or replacing any such vegetation.
- 4.23 <u>Alterations.</u> No Owner shall make any alterations or modifications to the exterior of the buildings, fences, railings or walls of the Project's structure without the prior written consent of the ARB. Any structural, plumbing or electrical modification, alteration or addition to or of a Unit shall (a) conform to the standards for construction contained in the Santa Cruz Building Code or, if applicable, the California Administrative Code, as amended from time to time, and (b) be approved, in advance, by the ARB.
- **4.24** Effect on Insurance. No Owner shall permit anything to be done or kept in his or her Unit or in the Common Areas which will result in any increase in the Association's insurance premiums or in the cancellation of insurance on any part of the Project or which would violate any law or the Common Areas Lease.
- <u>Delegation of Use</u>. Any Owner may delegate his or her rights of use and enjoyment of the Project, including any recreational facilities, to the members of his or her family, servants, guests, tenants (to the extent permitted hereby), and invitees, and to such other persons as may be permitted by the Bylaws and the Association Rules and Regulations, subject however, to this Declaration. Nevertheless, if an Owner has leased or rented a Unit, the Owner, and members of the Owner's family, servants, guests, tenants and invitees shall not be entitled to use and enjoy any of such rights in the Project while the Owner's Unit is occupied by the tenant. Instead, the tenant, while occupying such Unit, shall be entitled to use and enjoy such rights, including any recreational facilities, and can delegate the rights of use and enjoyment in the same manner as if tenant were an Owner during the period of his or her occupancy. Each Owner shall notify the secretary of the Association within five (5) calendar days after the entry into a lease for a Condominium of the period for which the lease will be effective and of the names of each tenant of such Owner's unit. Each Owner or tenant shall also notify the secretary of the Association of the names of all persons to whom such Owner or tenant has delegated any rights of use and enjoyment and the relationship that each such person bears to the Owner or tenant. Any delegated rights of use and enjoyment are subject to suspension to the same extent as are the rights of

Owners. No such delegation shall relieve an Owner from liability to the Association or to other Owners for payment of assessments or performance of the covenants, conditions, and restrictions contained in this Declaration.

- **4.26** <u>Single Family Use.</u> No Unit shall be occupied by more than a Single Family as defined herein.
- 4.27 Indemnification. Each Owner shall be liable to the remaining Owners for any damage to the Common Areas or to Association-owned property resulting from any intentional act or negligence of that Owner, that Owner's family, servants, guests, tenants, or invitees, but only to the extent that such damage is not covered by casualty insurance in favor of the Association. Each Owner, by accepting a Condominium deed, agrees to indemnify each and every other Owner (the "Indemnitees"), and to hold each Indemnitee harmless from, and to defend him or her against any claim of any person for personal injury or property damage occurring within the Unit of the indemnifying Owner, including Exclusive Use Common Areas, except to the extent that (i) such injury or damage is compensated by liability insurance in favor of the Association or the Indemnitee, or (ii) the injury or damage occurred by reason of the willful or negligent act or omission of the Association or the Indemnitee.
- 4.28 Enforcement and Remedies. The objective of this Declaration shall be to promote and seek voluntary compliance by Owners with the environmental standards and property use restrictions contained herein rather than the use of the sanctions in Section 7.6. Accordingly, in the event that the Association becomes aware of an architectural or property use infraction that does not necessitate immediate corrective action under Section 7.6 hereof, the Owner responsible for the violation shall receive written notice thereof and shall be given a reasonable opportunity to comply voluntarily with the pertinent provision(s) of this Declaration. Such notice shall describe the noncomplying condition, request that the Owner correct the condition within a reasonable time specified in the notice, and advise the Owner of his or her appeal rights. The failure of any Owner to comply with any provision of this Article 4, or with an order or decision of the Association, shall result in creation of a cause of action of the Association, and of any aggrieved Owner for the recovery of damages or for injunctive relief or both.

ARTICLE 5

MAINTENANCE AND ALTERATIONS

5.1 Owner's Responsibilities.

- (a) Each Owner of a Condominium shall be responsible for repairing and maintaining his or her Unit, including the equipment and fixtures in the Unit and the interior walls, ceilings, windows, and doors of the Unit, and shall maintain his or her Unit in a clean, sanitary, workable, and attractive condition. However, each Owner has complete discretion as to the choice of furniture, furnishings, and interior decorating, except that windows can be covered only by drapes, shutters, blinds or shades and cannot be painted or covered by foil, cardboard, or other similar materials. Each Owner also shall be responsible for repair and replacement of all interior glass, for repair, replacement and interior and exterior cleaning of the windows, and repair, replacement and interior cleaning of skylights of his or her Unit.
- (b) Each Owner shall, at his or her own expense, clean, maintain, and repair any Exclusive Use Common Areas over which the Owner has an easement. In any case in which two or more Owners have easements to use and enjoy the same Exclusive Use Common Areas, such Owners will share equally in expenses incurred for such cleaning, maintenance and repair.
- **5.2** <u>Association's Responsibilities.</u> Subject to the provisions of the Common Areas Lease, and as set forth more fully in Section 7.9 of this Declaration, the Association shall be responsible for repairing and maintaining the Common Areas in a clean, sanitary, workable, and attractive condition.

ARTICLE 6

TAXES AND ASSESSMENTS

6.1 Separate Assessments. To the extent allowed by law, all Condominiums in the Project, including their pro rata undivided interests in all the Common Areas and the membership of an Owner in the Association, shall be separately assessed and taxed so that all taxes, assessments, payments-in-lieu of taxes, and charges that may become liens prior to First Mortgages under local law shall relate only to the individual Condominium and not to the Project as a whole. Each Owner shall be obligated to pay any such taxes or assessments against his or her Condominium and against his or her personal property. In the event, however, that any Owner fails to pay any such taxes or assessments, the Association may make payment of them, provided that any such Owner shall promptly repay to the Association the full amount of such taxes or assessments, including amounts required to compensate the Association for its administrative costs and interest at the rate of ten percent (10%) per annum or at such other rate as the Board may lawfully impose from time to time. Nothing herein shall

prevent any Owner from contesting the imposition of any tax or the amount of the assessment of his or her Unit.

6.2 Payment Prior to Segregation. If real property taxes or payments-in-lieu of taxes are to be imposed on the Project, until such time as such taxes or payments-in-lieu of taxes have been segregated by the Santa Cruz County Assessor, they shall be paid by the respective Owners. The proportionate share of the taxes or payments for a particular Condominium shall equal the Percentage Interest of a Condominium. If, and to the extent that, any taxes are not paid by any Owner of a Condominium and are allowed to become delinquent, they may be paid by the Association and shall be collected from the delinquent Owner by the Association as provided in Section 6.1, above.

ARTICLE 7

THE ASSOCIATION

- 7.1 <u>Incorporation.</u> The Association shall be a nonprofit mutual benefit corporation formed under the laws of the State of California. At the recording of the first Condominium sale to an Owner other than Declarant, the Association shall be charged with the duties and invested with the powers set forth in the Articles, the Bylaws, and this Declaration.
- 7.2 Action By Board, Officers, and Designees. Except as to matters requiring the approval of Owners as set forth in this Declaration, the Articles, or the Bylaws, the affairs of the Association, including the exercise of its powers and duties, shall be conducted by the Board, such officers as the Board may elect or appoint, or such persons or entities with delegated authority under the provisions of Section 7.7 of this Declaration.
- 7.3 General Statement of Powers. The Association shall have all the powers of a nonprofit mutual benefit corporation organized under the General Nonprofit Mutual Benefit Corporation Law of California, as now in effect or as hereinafter amended or replaced, subject only to such limitations on the exercise of its powers as are set forth in the Articles, the Bylaws, and this Declaration. It shall have the power to do any lawful thing that may be authorized, required, or permitted to be done by the Association under this Declaration, the Articles, and the Bylaws, and to do and perform any act that may be necessary or proper for or incidental to, the exercise of any of the express powers of the Association, including without limitation, the acts enumerated in Section 7.4 through 7.7 of this Declaration, and the right to be sued by an Owner.

- 7.4 <u>Rights of Assessment.</u> The Association may establish, fix, and levy assessments against the Owners and collect and enforce payment of such assessments, in accordance with the provisions of this Declaration.
- 7.5 Right To Establish Rules. The Association may, subject to the retained rights of the Ground Lessor, adopt, amend, and repeal Association Rules and Regulations as it considers appropriate. The Association Rules and Regulations shall regulate the use and enjoyment of the Common Areas. A copy of the Association Rules and Regulations as adopted, amended, or repealed shall be mailed or otherwise delivered to each Owner and a copy shall be posted in a conspicuous place within the Project. If any provision of this Declaration, the Articles, the Bylaws, or the Common Areas Lease, is inconsistent with or materially alters any Association rule or regulation, the Declaration, the Articles, the Bylaws, or the Common Areas Lease, shall control to the extent of any such inconsistency.

7.6 Right To Impose Sanctions.

- (a) In addition to any other enforcement rights described in this Declaration and the Bylaws, or authorized by law, and subject to any restrictions on the Association's enforcement rights, including any due process requirements, imposed by this Declaration, the Bylaws, or by law, the Association may take any of the following actions against any person or entity whose act or failure to act violates or threatens to violate any provision of this Declaration, the Bylaws, the Common Areas Lease, or the Association Rules and Regulations:
 - (i) Impose monetary penalties, including late charges and interest.
 - (ii) Suspend voting rights in the Association.
 - (iii) Suspend use privileges for the Common Areas.
- (iv) Commence a legal action for declaratory judgment, damages, or injunctive relief, or any combination of the foregoing.
- (b) The determination whether to impose any of the sanctions set forth in Section 7.6(a) shall be within the sole discretion of the Association. Any legal action may be brought in the name of the Association on its own behalf and on behalf of any Owner who consents, and the prevailing party in any such action shall be entitled to recover costs and reasonable attorneys' fees.

- (c) The Association may take more than one of the foregoing enforcement actions against any one violation or threatened violation, provided that a suspension of use privileges shall not exceed thirty (30) calendar days (unless the suspension is for delinquent assessments) and no monetary penalty shall exceed that permitted by to section 1366.1 of the California Civil Code, as it shall hereafter be amended.
- (d) The Association, in its sole discretion, may resolve or settle any dispute to which it is a party, including any legal action, on its own behalf and under such terms and conditions as it considers appropriate.
- (e) The enforcement of monetary penalties is subject to the restrictions described in Section 9.7 of this Declaration.
- (f) If an Owner fails to cure a default or any violation of the Condominium Instruments within sixty (60) calendar days after written notice to that Owner, the Association shall give the notice required in Section 15.9 of this Declaration.
- 7.7 <u>Right of Delegation.</u> The Association may delegate any of its powers and duties to its officers, employees, committees, or agents, including The Regents of the University of California, or a professional management agent.
- 7.8 <u>General Statement of Duties.</u> In addition to any duties described in the Articles or Bylaws, the Association shall have the duties set forth in Sections 7.9 through 7.15 of this Declaration.
- Areas. Subject to the provisions of the Common Areas Lease, the Association shall manage and maintain in good condition and repair the Common Areas, including the facilities, improvements, landscaping, and other personal and real property acquired by or subject to the control of the Association, and shall clean, maintain and repair all Exclusive Use Common Areas, if not properly cleaned, maintained or repaired by an Owner pursuant to Section 5.1(b) of this Declaration. The Association shall create and maintain a reserve fund out of regular assessments for the replacement of all elements of the Common Areas. Subject to the provisions of Section 9.7(c) of this Declaration, the Association may levy a special assessment against any Owner who has an easement to use any Exclusive Use Common Areas on which the Association performs such cleaning, maintenance or repair to reimburse the Association for any costs incurred therein.

7.10 Contracts for Goods and Services.

- (a) Except as expressly limited by this Declaration, the Association shall enter into such contracts for services or materials as may be necessary to perform its duties, including contracts with Declarant.
- (b) Any agreement between the Association and Declarant pursuant to which Declarant agrees to provide services, and any agreement for professional management by any other manager, shall provide for termination by either party without cause or payment of a termination fee on thirty (30) calendar days' written notice and shall have a maximum contract term of one (1) year, provided that the Association can renew any such contract on a year-to-year basis. If the Project is professionally managed, the Association shall not terminate professional management and assume self-management without the consent of sixty-seven percent (67%) of the voting rights of Owners, or of each class of Owners if a two-class voting system is in effect, and of fifty-one percent (51%) of First Mortgagees.
- 7.11 Payment of Taxes and Assessments. If taxes or assessments authorized by law are levied against the Common Areas or any property owned by the Association, the Association shall pay such taxes or assessments. Such taxes and assessments may be contested or compromised by the Association, provided that they are paid or that a bond insuring payment is posted before the sale or the disposition of any property to satisfy the payment of such taxes.
- **7.12** Furnishing of Utilities. The Association shall acquire, provide, and pay for water, sewer, garbage disposal, refuse and rubbish collection, electrical, telephone, gas, and other necessary utility services for the Common Areas and the Condominiums to the extent that such services are not separately billed.
- 7.13 <u>Securing Insurance</u>. The Association shall obtain and maintain the insurance described in Article 10 of this Declaration.
- 7.14 <u>Preparation and Distribution of Financial Information and Governing Instruments.</u> The Association shall prepare and distribute the following financial statements, reports, and copies of the governing instruments as indicated:
 - (i) A pro forma operating budget for each fiscal year, including:
 - (A) estimated revenues and expenses on an accrual basis;
 - (B) identification of the total cash reserves currently set aside;

- (C) identification of the estimated remaining life and the methods of funding used to defray the future repair, replacement, or additions to major components for which the Association is responsible; and
- (D) a general statement addressing the procedures used for the calculation and establishment of reserves to defray the future repair, replacement, or additions to major components for which the Association is responsible.

A copy of the pro forma operating budget shall be distributed to each Owner and to any Mortgagee, if such Mortgagee has requested a copy, not less than forty-five (45) nor more than sixty (60) calendar days prior to the beginning of each fiscal year.

- (ii) An annual report consisting of a balance sheet rendered as of the last day of the fiscal year, an operating statement for the fiscal year, and a statement of changes in financial position for the fiscal year. A copy of the annual report shall be distributed to each Owner and to any Mortgagee, if such Mortgagee has requested a copy, within one hundred twenty (120) calendar days after the close of the fiscal year. So long as there are more than fifty (50) units in the Project or in any fiscal year in which the gross income of the Association exceeds the amount stated in section 1365(b) of the California Civil Code, a review of the annual report shall be prepared by a licensee of the California State Board of Accountancy in accordance with generally accepted accounting principles. If the annual report is not reviewed by an independent accountant, the report shall be accompanied by the certificate of an authorized officer of the Association that the report was prepared from the books and records of the Association without independent audit or review.
- (iii) A statement of the Association's policies and practices in enforcing its remedies against Owners for delinquent regular or special assessments, including the recording and foreclosing of liens against a delinquent Owner's Condominium. A copy of this statement shall be distributed to each Owner and to any Mortgagee that has requested a copy within sixty (60) calendar days prior to the beginning of each fiscal year.
- (iv) Copies of this Declaration, the Articles, the Bylaws, the Association Rules and Regulations, the most recent annual report and other financial materials distributed pursuant to Section 7.14(ii) above, and the statement regarding delinquent assessments as described in Section 9.11 of this Declaration to any Owner within ten (10) calendar days of the mailing or delivery of a written request. The Board may impose a fee to provide these materials that is not to exceed the Association's reasonable costs in preparing and reproducing the materials.

- 7.15 Other Duties. The Association shall perform such other acts as may be reasonably necessary to exercise its powers or perform its duties under any of the provisions of this Declaration, the Articles, the Bylaws, the Common Areas Lease, the Association Rules and Regulations, or Board resolutions.
- 7.16 <u>Limitations on Authority of Board and the Association.</u> Except with the vote or written consent of the Owners holding fifty-one percent (51%) of the voting rights of each class of Owners, if two classes exist, or, if only one class exists, fifty-one percent (51%) of the voting rights of Owners (including Declarant) and fifty-one percent (51%) of the voting rights of Owners other than Declarant, the Board shall not take any of the following actions:
- (i) Except as permitted by law, incur aggregate expenditures for capital improvements other than budgeted and reported capital expenditures to the Common Areas in any fiscal year in excess of fifteen percent (15%) of the budgeted gross expenses of the Association (including budgeted capital expenses) for that fiscal year.
- (ii) Sell during any fiscal year property of the Association having an aggregate fair market value greater than five percent (5%) of the budgeted gross expenses of the Association for that fiscal year.
- (iii) Pay compensation to members of the Board of Directors or to officers of the Association for services performed in the conduct of the Association's business, provided that the Board may reimburse expenses incurred in carrying on the business of the Association.
- (iv) Enter into a contract with a third person to furnish goods or services for the Common Areas or the Association for a term longer than one (1) year, with the following exception: enter into any contracts which bind it or the Association for a period in excess of one (1) year, provided that any management agreement for the Project shall further provide that it can be cancelled upon thirty (30) calendar days' written notice by the Association, except for the following:
- (1) a management contract, the terms of which have been approved by the Federal Housing Administration or Veteran's Administration;
- (2) the contract of a public utility company if the rates charged for the materials or services are regulated by the California Public Utilities Commission;

provided, however, that the term of the contract shall not exceed the shortest term for which the supplier will contract for the regulated rate;

- (3) prepaid casualty and/or liability insurance policies for a term not to exceed three (3) years' duration provided that the policy permits a short rate cancellation by the insured;
- (4) lease agreements for fixtures and equipment, which shall not exceed five (5) years' duration, provided that the Lessor under the agreement is not an entity in which the Declarant has a direct or indirect ownership interest of ten percent (10%) or more; or
- (5) agreements for cable or microwave communication services and equipment, which shall not exceed five (5) years' duration, provided that the supplier is not an entity in which the Declarant has a direct or indirect ownership interest of ten percent (10%) or more.
- 7.17 <u>Limitation on Liability of Directors and Officers.</u> No director, officer, committee member, employee, or other agent of the Association, including Declarant or any agent of Declarant when acting in such capacity, shall be liable to any Owner or any other party, including the Association, for any damage, loss, or prejudice suffered or claimed on account of any act, omission, error, or negligence of any such person if such person is found, pursuant to the procedures set forth in the Bylaws, to have complied with the standards of care as set forth in the Bylaws or in the relevant provisions of the California Corporations Code if more stringent than those set forth in the Bylaws, or to otherwise be entitled to be indemnified pursuant to the Bylaws.

7.18 <u>Inspection of Books and Records.</u>

(a) Any Owner, or that Owner's duly appointed representative, shall have access to the Association's membership register, books of account, and minutes from any meeting of the Owners, the Board, or any committee of the Board in order to inspect and copy such records for any purpose reasonably related to his or her interest as an Owner. Any Owner may, at his or her own expense, cause an audit or inspection to be made of the books and financial records of the Association. Access shall be at any reasonable time at the office of the Association or such other place within the area of the City of Santa Cruz as the Board prescribes. The Board shall establish rules regarding the notice the Owner must give to the custodian of the records or the Board to obtain access, the hours and days of the week when the records may be inspected and copied,

and the charges to be imposed by the Association for copying records requested by the Owner.

- (b) Any member of the Board may at any reasonable time inspect, copy, or make extracts of any books, records, or documents of the Association and inspect for physical properties owned or controlled by the Association.
- (c) The provisions of this Section 7.18 shall be further governed by sections 8330, 8333, and 8334 of the California Corporations Code, as now in effect or as hereinafter amended or replaced.

ARTICLE 8

ASSOCIATION MEMBERSHIP AND VOTING

- 8.1 <u>Membership Appurtenant to Ownership.</u>
- (a) Each Owner, including Declarant, shall be a Member of the Association. Membership shall be appurtenant to each Condominium in the Project, and the holding of an ownership interest in a Condominium in the Project shall be the sole qualification for membership, provided that no Owner shall hold more than one membership even though such Owner owns an interest in more than one Condominium. Membership shall terminate automatically when the Owner no longer holds any ownership interest in any Condominium in the Project. Membership may not be assigned, transferred, pledged, hypothecated, conveyed, or alienated in any way except on a transfer of title to a Condominium and then only to the transferee. Any attempt to make a prohibited transfer shall be void. Any transfer of title to a Condominium shall transfer automatically the appurtenant membership to the transferee. Any party that holds an interest in a Condominium merely as security for performance of an obligation shall not be a Member of the Association.
- (b) Each Member of the Association shall have the rights, duties, and obligations set forth in this Declaration, the Articles, the Bylaws, and the Association Rules and Regulations.
- (c) Notwithstanding any other provision of this Declaration, the Articles, or the Bylaws, throughout the Declarant Control Period, Declarant shall be deemed to be the Class B Owner in the Association, as defined in Section 8.3(a) below, regardless of the number of Condominiums that it owns at any point during that Period.

8.2 Approval by a Specified Percentage of Owners. Except as required by law or as otherwise provided in this Declaration, the Articles, or the Bylaws, and subject to the provisions of Section 8.3 of this Declaration, all matters requiring the approval of a majority or a specified percentage of Owners or of the voting rights or voting power of Owners shall be deemed approved if approved at any duly called regular or special meeting at which a quorum is present, either in person or by proxy, by Owners holding the specified percentage of the total voting power of Owners present, either in person or by proxy, or if approved by written ballot in the absence of a meeting, in accordance with provisions, if any, of the Bylaws duly permitting the use of a written ballot.

8.3 Voting.

- (a) The Association shall have two classes of voting membership as follows:
- (i) Class A: Class A Owners are all Owners, with the exception of Declarant. Each Class A Owner shall be entitled to one vote for each Condominium in which he or she owns an interest. If more than one Owner owns an interest in the Condominium, only one vote may be cast with respect to that Condominium.
- (ii) Class B: The Class B Owner shall be Declarant, who shall be entitled to three votes for each Condominium it owns or has the right to create in any phase of the Project pursuant to Section 17.2 of this Declaration. Class B membership shall cease and be converted to Class A membership on the expiration of the Declarant Control Period.
- (b) As long as two classes of voting memberships exist, except as expressly provided otherwise in this Declaration, any action by the Association that requires approval by the Owners or by the voting rights or voting power of a specified percentage of Owners shall require the approval by the designated percentage of voting power in each class. Except as required by law, or as otherwise provided in this Declaration, the Articles or the Bylaws, matters which require the approval or vote of Class A Owners will be deemed approved by Class A Owners if approved by the specified percentage of Class A Owners present, either in person or by proxy, at a duly called regular or special meeting at which at least thirty-three and one-third percent (33-1/3%) of the voting power of Class A Owners is present, either in person or by proxy, or if approved by written ballot by the specified percentage of Class A Owners if the number of such votes cast by ballot equals or exceeds thirty-three and one-third percent (33-1/3%) of the voting power of Class A Owners.
 - (c) Voting rights shall vest at the time of conveyance of any Condominium.

- (d) Voting for the members of the Board of Directors shall be by secret written ballot, and all Owners, excluding the Class B Owner during the Declarant Control Period, shall be entitled to exercise cumulative voting rights as provided in Section 8.5 of this Declaration.
- 8.4 <u>Joint Ownership Votes.</u> The vote that is attributed to each Condominium may not be cast on a fractional basis. If a Condominium has more than one Owner and the Owners are unable to agree as to how the vote should be cast, the vote shall be forfeited on the matter in question. If one Owner casts the vote attributed to a Condominium, the vote shall conclusively bind all the Owners of that Condominium. If more than one Owner casts the vote attributed to a Condominium in any matter in which only one vote can be cast for that Condominium, the votes cast by such Owners shall not be counted and shall be considered void.

8.5 <u>Cumulative Voting.</u>

- (a) The election of members to the Board of Directors may be by cumulative voting as described herein, provided that a candidate's name has been placed in nomination prior to the voting and at least one (1) Owner has given notice at the meeting prior to the voting of his or her intention to cumulate votes. If any such Owner has given such notice, then all Owners shall have the right to cumulate their votes for candidates in nomination.
- (b) Under cumulative voting, each Owner, either in person or by proxy, may give a single candidate the number of votes equal to the number of Directors for whom the Owner is entitled to vote multiplied by the number of votes the Owner is entitled to exercise under this Declaration, or the Owner may distribute these cumulated votes among any two or more candidates as the Owner desires. Subject to the provisions of the Bylaws, the candidates receiving the highest number of votes, up to the number of Directors to be elected, shall be elected.
- (c) Unless the entire Board is removed by a vote of the Owners, an individual director may not be removed prior to the expiration of his or her term if the votes against removal would have been sufficient to elect that director if cast cumulatively at an election at which the same total number of votes were cast and all Directors authorized at the time of the most recent election of that director were being elected, except that this provision shall not apply to Declarant during the Declarant Control Period.

- (d) These cumulative voting provisions may not be applied to deny Owners other than Declarant the right, as set forth in the Bylaws, to elect two of the Directors during the Declarant Control Period.
- (e) Notwithstanding the foregoing provisions of this Section 8.5, the Class B Owner shall not be entitled to exercise cumulative voting rights during the Declarant Control Period.

ARTICLE 9

ASSESSMENTS

9.1 Payment of Assessments.

- (a) Declarant covenants and agrees for each Condominium owned by it in the Project, and each Owner by acceptance of a deed, covenants and agrees for each Unit owned, to pay to the Association the initial working capital payment and the regular and special assessments levied in accordance with the provisions of this Declaration, and to allow the Association to enforce any assessment lien established in accordance with the provisions of this Declaration by nonjudicial proceedings under a power of sale or by any other means authorized by law.
- (b) Notwithstanding the provisions of Section 9.1(a), Declarant reserves the right, for a period of up to one (1) year following the date of conveyance of the first Condominium, to pay both the operating expenses of the Association and funds necessary to establish adequate reserves in lieu of imposition of assessments by the Association.

9.2 <u>Initial Working Capital Payment.</u>

(a) Each Condominium in the Project shall be liable for an initial working capital payment in an amount equal to two (2) months' regular assessment (as estimated by Declarant at the time of sale). Such payment is in addition to any assessments imposed pursuant to this Declaration. Declarant, as the agent of the Board of Directors, will collect such payment from each initial purchaser other than Declarant at the time of closing and will deliver the funds collected to the Board of Directors for deposit in a segregated account. Such funds may be used for prepaid expenses, initial equipment, supplies, organizational costs, other start-up costs, and for such other purposes as the Board may determine.

- (b) If a deed transferring title to the last Condominium to an Owner other than Declarant has not been recorded within sixty (60) calendar days of the recordation of a deed conveying title to the first Condominium sold, Declarant shall pay the initial working capital payment to the Association for each unsold Condominium. Declarant shall be entitled to reimbursement of such payment at the time of recordation of a deed conveying title to each Condominium for which Declarant made the payment.
- 9.3 <u>Limited Exemption During Construction.</u> Notwithstanding the provisions of Section 9.1 of this Declaration, Declarant or any other Owner shall not be obligated to pay any portion of a regular or special assessment that is for the purpose of defraying expenses and establishing reserves directly attributable to the existence and the use of any Condominium until occupancy for the Unit of that Condominium has been authorized by the State Fire Marshall or his or her delegate. This assessment exemption includes, but is not limited to, assessments levied for the following purposes: roof replacement, exterior maintenance, walkway and common area lighting, refuse disposal, cable television, and domestic water supplied to occupied Units.
- 9.4 Personal Obligation of Owner. Each assessment or installment, together with any late charge, interest, collection costs, and reasonable attorneys' fees, shall be the personal obligation of the Owner at the time such assessment or installment becomes due and payable. If there is more than one Owner of a particular Condominium, each Owner shall be jointly and severally liable. The personal obligation for any delinquent assessments or installments and related sums shall not pass to an Owner's successor in interest unless expressly assumed by that successor in interest, provided that any lien held by the Association, if properly recorded in accordance with Section 9.13 of this Declaration, shall not be affected by this provision. No Owner may be relieved from the obligation to pay assessments or installments by waiving the use or enjoyment of all or any portion of the Common Areas or the Owner's Condominium, or by abandoning the Condominium.

9.5 Scope of Assessment Authority.

(a) The assessments levied by the Association shall be used exclusively to promote the recreation, health, safety, and welfare of the members of the Association; to improve, replace, repair, operate, and maintain the Common Areas and the improvements and personal property in the Common Areas that are owned, leased or maintained by the Association; to provide funds necessary for the performance of the duties of the Association as set forth in this Declaration; and to further any other purpose that is for the common benefit of the Owners in their use and enjoyment of the Project.

(b) The Association shall establish and maintain a reserve fund adequate for it to perform its responsibilities, as set forth in this Declaration, the Articles or the Bylaws, with regard to the periodic maintenance, repair and replacement of improvements to the Common Areas. Such reserve fund shall be maintained out of regular assessments for common expenses.

9.6 Regular Assessments.

- (a) Not more than ninety (90) calendar days nor less than sixty (60) calendar days before the beginning of each fiscal year of the Association, the Board, without the requirement for a vote of the Owners, shall establish the regular assessment for the forthcoming fiscal year.
- (b) Except as permitted by law, no regular assessment shall be set in an amount that is more than ten percent (10%) greater than the amount of the regular assessment for the preceding fiscal year without the consent of a majority of the voting rights of Owners, or of the voting rights of each class of Owners if a two-class voting system is in effect.
- (c) Unless the Association is exempt from federal or state income taxes, including, without limitation, an exemption under section 528 of the Internal Revenue Code and section 23701t of the California Revenue and Taxation Code, as now in effect or as hereinafter amended or replaced, all reserve funds, to the extent possible, shall be designated and accounted for as capital contributions to the Association, and the Board shall take such steps as may be reasonably necessary under federal and state tax laws to prevent the reserve funds from being taxed as income of the Association, including, if necessary, maintaining the reserve funds in segregated accounts and not commingling the fund with general operating funds.

9.7 Special Assessments.

(a) The Board may levy a special assessment if the Board, in its discretion, determines that the Association's available funds are or will become inadequate to meet the estimated expenses of the Association, including the maintenance of appropriate reserves, for a particular fiscal year for any reason, including, but not limited to, unanticipated delinquencies, costs of construction, unexpected repairs or replacements of capital improvements or otherwise. The Board shall determine the amount necessary to meet the estimated expenses, and if the amount is approved by a majority vote of the Board, it shall become a special assessment, except that the Board may not, except as permitted by law, impose special assessments which in the aggregate exceed five percent (5%) of the budgeted gross expenses of the Association (including budgeted capital expenses) for that fiscal year without the consent of a majority of the voting rights of Owners or of the voting rights of Owners if a two-class voting

system is in effect. The Board, in its discretion, may levy the entire assessment immediately or levy it in installments over a period it considers appropriate.

- (b) Unless the Association is exempt from federal or state income taxes, including without limitation an exemption under section 528 of the Internal Revenue Code and section 23701t of the California Revenue and Taxation Code, as now in effect or as hereinafter amended or replaced, the Board shall take such steps as may be reasonably necessary to prevent the special assessment from being included in the Association's income for federal and state income tax purposes, including, if necessary, depositing funds in a segregated account, not commingling the funds with any other funds of the Association, and using the funds solely for the purpose for which they were levied.
- After compliance with the due process requirements in the Bylaws, the Board may impose a monetary penalty and levy a special assessment against a particular Condominium to reimburse the Association for any taxes, assessments or other expenses paid by the Association on behalf of the Owner(s) of the Condominium assessed; costs incurred in repairing damage to the Common Areas, or any improvements or personal property located thereon, for which the Owner(s) of the Condominium assessed was allegedly responsible; costs incurred in cleaning, maintaining or repairing any Exclusive Use Common Areas not properly cared for by an Owner pursuant to Section 5.1(b) of this Declaration; or costs incurred in bringing the Owner or the Owner's Condominium into compliance with this Declaration, the Articles, the Bylaws, or the Association Rules and Regulations, provided that, except as permitted by law, a special assessment related to a monetary penalty may not become a lien against the responsible Owner's Condominium that is enforceable by a power of sale under sections 2924, 2924b and 2924c of the California Civil Code. This restriction on enforcement is not applicable to late payment penalties for delinquent assessments or to charges imposed to reimburse the Association for loss of interest or for collection costs, including reasonable attorneys' fees, for delinquent assessments.
- 9.8 <u>Allocation of Assessments.</u> Subject to the provisions of Section 9.4 of this Declaration, the regular and special assessments levied by the Board shall be allocated in accordance with the Percentage Interest of each Unit, except that special assessments levied against a particular Condominium pursuant to Section 9.7(c) of this Declaration shall not be subject to this allocation provision.
- 9.9 <u>Fiscal Year and Assessment Period.</u> Unless the Board determines otherwise, the Association's fiscal year shall commence on July 1 of each year and shall terminate on June 30 of the next year, provided that the first regular assessment period for all Condominiums in any phase shall commence on the first day of the calendar month following the date of the closing of

the first conveyance of a Condominium, and shall terminate on June 30 of that year. The regular assessment shall be payable in equal monthly installments unless the Board adopts some other method for payment. The first regular assessment and all special assessments shall be adjusted according to the number of months remaining in the fiscal year and shall be payable in equal monthly installments unless the Board adopts some other method for payment.

9.10 <u>Due Date, Late Charges, and Interest.</u>

- (a) At least ten (10) calendar days prior to the date on which any regular or special assessment is due, the Board shall give each Owner written notice of the amount of assessment and the due date (or if paid in installments, the due dates and the amount of each installment). The notice need be given only once, at the beginning of each fiscal year for any regular assessment paid in installments, and once for any special assessment paid in installments. Unless the Board specifies otherwise, the installment due dates shall be the first day of each month.
- (b) Any assessment payment, including any installment payment, shall become delinquent if payment is not received by the Association within fifteen (15) calendar days after its due date. There will be a late charge of Ten Dollars (\$10.00) or ten percent (10%) of the delinquent assessment payment, if greater, or at the option of the Board, such higher amount on each delinquent payment as may be permitted by law. A late charge may not be imposed more than once on any delinquent payment, shall not eliminate or supersede any charges imposed on prior delinquent payments, and shall constitute full compensation to the Association for any additional bookkeeping, billing, or other administrative costs resulting from the delinquent payment. Nothing in this provision shall affect the right of the Association to recover its costs of collection and enforcement, as well as reasonable attorneys' fees, if it prevails in an action brought to collect delinquent payments pursuant to the provisions of Section 9.12 of this Declaration.
- (c) Interest shall also accrue on any delinquent payment at the rate of ten percent (10%) per annum or at such rate as the Board may lawfully impose from time to time. Interest shall accrue from a date thirty (30) calendar days after the due date of the assessment through and including the date full payment is received by the Association.

9.11 Estoppel Certificates.

(a) Within ten (10) calendar days after receipt of a written request by an Owner, the Board shall provide the Owner with a written statement containing the following information: (i) whether, to the knowledge of the Association, the Owner or the Owner's Condominium is in violation of any of the provisions of this Declaration,

the Articles, the Bylaws, or the Association Rules and Regulations; (ii) the amount of regular and special assessments, including installment payments, paid by the Owner during the fiscal year in which the request is received; and (iii) the amount of any delinquent assessments, penalties, interest, attorneys' fees, and other charges on the Owner's Condominium as provided by this Declaration, the Articles, the Bylaws, or the Association Rules and Regulations.

- (b) The Board may charge the Owner a fee to recover its reasonable costs in preparing the statement.
- (c) Any prospective purchaser or Mortgagee of the Owner's Condominium may rely on the information in this written statement, provided that reliance may not extend to any violation of the Declaration, the Articles, the Bylaws, or the Association Rules and Regulations of which the Association does not have actual knowledge.
- enforce assessments. In addition to the enforcement powers described in Section 7.6 of this Declaration, and subject to the restriction on the enforcement of monetary penalties described in Section 9.7 of this Declaration, the Association may enforce delinquent assessments, including delinquent installments, by suing the Owner directly on the debt established by the assessment, or by establishing a lien against the Owner's Condominium as provided in Section 9.13 of this Declaration and foreclosing the lien through either judicial proceedings or nonjudicial proceedings under a power of sale as provided in Section 9.14 of this Declaration. The Association may commence and maintain a lawsuit directly on the debt without waiving its right to establish a lien against the Owner's Condominium for the delinquent assessment. In any action instituted by the Association to collect delinquent assessments, accompanying late charges, or interest, the prevailing party shall be entitled to recover costs and reasonable attorneys' fees.

9.13 Creation of Assessment Lien.

(a) A delinquent assessment or installment, together with any accompanying late charges, interest, costs (including reasonable attorneys' fees), and penalties as may be authorized under this Declaration shall become a lien on the Condominium against which the assessment was levied upon recordation in the Official Records of the County Recorder of Santa Cruz County, California. The notice shall describe the amount of the delinquent assessment or installment, the related charges authorized by this Declaration, a description of the Condominium, the name of the Owner, and the name and address of any trustee authorized by the Association to conduct a foreclosure sale,

and shall be signed by any officer of the Association, or by any employee or agent of the Association authorized to do so by the Board.

- (b) Unless the Board considers the immediate recording of the notice to be in the best interest of the Association, the notice shall not be recorded until at least fifteen (15) calendar days after the Association has delivered to the delinquent Owner a written notice of default and a demand for payment, provided that payment has not been made prior to recordation of the notice. If the delinquent assessment or installment and related charges are paid or otherwise satisfied, the Association shall record a notice of satisfaction and release of lien.
- (c) Within thirty (30) calendar days after the Association has delivered to a delinquent Owner a written notice of default and a demand for payment, or within fifteen (15) calendar days after the date the Association records a lien on any Condominium against which a delinquent assessment was levied, whichever is sooner, the Association shall give to Declarant and to any Mortgagee who holds a Mortgage on the affected Unit and who has filed a request for notices with the Board of Directors pursuant to Section 15.2 of this Declaration, notice of the delinquency. Declarant, as well as any Mortgagee, shall have the right, but shall not be under any obligation, to pay such delinquent assessment and any costs, penalties, or interest assessed thereon, to avoid foreclosure of any lien under the provisions of Section 9.14 of this Declaration.
- <u>Foreclosure Under Assessment Lien.</u> Within two (2) years after the recording of a notice of assessment, but not less than fifteen (15) calendar days after the recording of the notice of assessment, the Board shall enforce any assessment lien established under Section 9.13 of this Declaration by filing an action for judicial foreclosure or by recording a notice of default in the form described in section 2924c(b)(1) of the California Civil Code to commence a nonjudicial foreclosure. Any nonjudicial foreclosure shall be conducted in accordance with the requirements of sections 2924, 2924b, 2924c, 2924f, 2924g, and 2924h of the California Civil Code, as now in effect or as hereinafter amended or replaced, that are applicable to nonjudicial foreclosures of Mortgages or deeds of trust, provided that the Association may, in accordance with the provisions of section 2924a of the California Civil Code, if applicable, appoint its attorney, any officer or director, or any title insurance company authorized to do business in California to conduct the sale. The Association may bid on the Condominium at the sale, and may hold, lease, mortgage, and convey the acquired Condominium, subject to compliance with the Common Areas Lease underlying such Condominium. If the default is cured before the sale, or before completing a judicial foreclosure, including payment of all costs and expenses incurred by the Association, the Association shall record a notice of satisfaction and release of lien and, on receipt of

a written request by the Owner, a notice of rescission rescinding the declaration of default and demand for sale.

- 9.15 <u>Waiver of Homestead Protection.</u> Each Owner, to the extent permitted by law, waives the protection of any declared homestead or homestead exemption under the laws of California as applied to any action to enforce the assessments levied by the Association.
- 9.16 <u>Common Areas Interest.</u> Following any sale or other transfer of ownership of a Condominium in conformity with the provisions of this Declaration, the Association shall, upon request of the new Owner, provide to such Owner, an instrument, in recordable form, evidencing such Owner's undivided leasehold interest in the Common Areas.

ARTICLE 10

INSURANCE

<u>Liability Insurance.</u> The Association shall obtain and maintain comprehensive public liability insurance insuring the Association, the Board, the Ground Lessor (unless waived thereby), any manager, Declarant, and the Owners and occupants of Condominiums, and their respective family members, guests, invitees, agents and employees, against any liability incident to the ownership or use of the Common Areas or any other Association-owned or maintained real or personal property, and including a cross-liability or severability of interest endorsement insuring each insured against liability to each other insured. The general aggregate limit of such insurance shall be not less than Two Million Current Dollars (\$2,000,000.00) (which amount shall be adjusted not less frequently than every third year after the date hereof, to reflect both inflationary changes and changes in the prudent level of insurance coverage) per person for personal or bodily injury, death or property damage and Two Million Current Dollars (\$2,000,000.00) (subject to adjustment as aforesaid) for any one accident or occurrence; and with limits of not less than One Million Current Dollars (\$1,000,000.00), as adjusted, per occurrence with respect to property damage. Such insurance shall also include coverage against water damage liability, liability for owned, non-owned and hired automobiles, liability for the property of others and against other liability or risk customarily covered with respect to projects similar in construction, location and use. Such insurance shall contain a severability of interest endorsement which shall preclude the insurer from denying the claim of named insureds because of any neglect or other act or omission of another named insured. At their respective option, the Declarant and the Ground Lessor shall be named as an additional insured with respect to any such insurance.

10.2 Fire and Casualty and Demolition Insurance.

- (a) The Association shall obtain and maintain a master or blanket policy of fire and casualty insurance and coverage for the full replacement insurable value of all of the improvements within the Project that covers one hundred percent (100%) of the current replacement cost of such improvements. The policy shall contain an "Agreed Amount" endorsement and a one hundred percent (100%) co-insurance clause and shall include coverage of all perils included in a standard "all risk" endorsement. The policy shall also contain, to the extent required by law or the Federal National Mortgage Association and/or the Federal Home Loan Mortgage Corporation, an increased cost of construction endorsement, and, to the extent available at reasonable cost, an inflation guard endorsement or its equivalent; vandalism and malicious mischief coverage; and a determinable cash adjustment clause or a similar clause to permit cash settlement covering full value of the improvements in case of partial destruction and a decision not to rebuild. The maximum deductible on such policy shall be the lesser of Ten Thousand Current Dollars (\$10,000.00) or one percent (1%) of the face amount of the policy, and the maximum deductible related to proceeds to be paid with regard to any individual Unit shall be the lesser of One Thousand Current Dollars (\$1,000.00) or one percent (1%) of the Unit's replacement cost.
- (b) The Association may obtain and maintain demolition insurance in adequate amounts to cover demolition, in case of total or partial destruction of the Project and a decision not to rebuild, and may obtain blanket policies of earthquake or flood insurance to the extent such coverage is customarily available to similar operations at reasonable cost, or is required by law or the Federal National Mortgage Association and/or the Federal Home Loan Mortgage Corporation.
- (c) The insurance policies required under this Section shall name as insured the Association, the Owners, and all Mortgagees as their respective interests may appear, and may contain loss payable endorsements in favor of the trustee described in Section 10.3 of this Declaration. In addition, the policies shall provide that the insurance coverage provided by the policies will not be prejudiced by any acts or omissions of individual Owners that are not under the control of the Association, and that the policies will be primary, even if an Owner has other insurance that covers the same loss. Further, the policies shall provide that any insurance trust agreement established pursuant to Section 10.3 of this Declaration will be recognized.
- 10.3 <u>Appointment of a Trustee.</u> All insurance proceeds payable to the Association under Section 10.2 of this Declaration for losses to real property and improvements, subject to the rights of Mortgagees under Section 10.7 of this

Declaration, may, at the direction of the Association, be paid to a trustee, to be held and expended for the benefit of the Owners, Mortgagees, and others, as their respective interests shall appear. The trustee may be the Association or a commercial bank or other financial institution with trust powers in the County of Santa Cruz, that agrees in writing to accept such trust.

10.4 Owner's Insurance.

- (a) Except as provided in this Section 10.4, no Owner shall separately insure his or her Unit against loss by fire or other casualty covered by any insurance carried under Section 10.2 of this Declaration. If any Owner violates this provision, any diminution in insurance proceeds otherwise payable under policies described in Section 10.2 of this Declaration that results from the existence of such other insurance will be chargeable to the Owner who acquired other insurance, and such Owner will be liable to the Association to the extent of any such diminution. An Owner may insure his or her personal property against loss. In addition, any improvements made by an Owner within his or her Unit may be separately insured by the Owner, provided that the insurance is to be limited to the type and nature of coverage commonly known as "tenant's improvements," and further provided that any improvements constructed with funds derived from a Mortgage on the Owner's Unit are to be covered by the insurance carried under Section 10.2 of this Declaration so long as such insurance is obtained. All such insurance that is individually carried must contain a waiver of subrogation rights by the insurer as to other Owners, the Association, Declarant, and other Institutional First Mortgagees of such Condominium, if such waiver is customarily obtainable.
- (b) An Owner may carry whatever personal liability and property damage liability insurance with respect to his or her Condominium he or she desires. However, any such policy shall include a waiver of subrogation clause acceptable to the Board and to any Institutional First Mortgagees, if such a waiver is customarily obtainable.

10.5 <u>Workers' Compensation and Other Association Insurance.</u>

- (a) The Association shall purchase and maintain workers' compensation insurance, to the extent that it is required by law, for all employees or uninsured contractors of the Association.
- (b) The Association shall purchase and maintain fidelity bonds or insurance or, if it has employed an outside management agent, will require such agent to obtain fidelity bonds or insurance. In either case, the Association shall be named as obligee in

such fidelity bonds or insurance. Such fidelity bonds or insurance shall be in an amount not less than twenty-five percent (25%) of each year's estimated annual operating expenses and reserves. In the case of fidelity bonds or insurance purchased by the Association, such bonds or insurance shall contain an endorsement of coverage of any person who may serve without compensation.

- (c) The insurance policies required under this Section shall name as insured the Association, the Owners, and all Mortgagees as their respective interests may appear, and may contain loss payable endorsements in favor of the trustee described in Section 10.3 of this Declaration. In addition, the policies shall provide that the insurance coverage provided by the policies will not be prejudiced by any acts or omissions of individual Owners that are not under the control of the Association, and that the policies will be primary, even if an Owner has other insurance that covers the same loss. Further, the policies shall provide that any insurance trust agreement established pursuant to Section 10.3 of this Declaration will be recognized.
- (d) The Association shall purchase and maintain insurance on personal property owned by the Association and may obtain and maintain any other insurance that it deems necessary.
- agree on the value and extent of any loss under any policy carried pursuant to the provisions of this Article 10. The Association is granted full right and authority to compromise and settle any claim or enforce any claim by legal action or otherwise and to execute releases in favor of any insurer. These provisions shall not apply to policies carried by an Owner pursuant to Section 10.4 of this Declaration.
- **10.7** <u>Distribution to Mortgagees.</u> Subject to Section 11.6 of this Declaration, any Mortgagee, pursuant to an agreement with its mortgagor, may apply insurance proceeds payable on account of a Condominium to reduce its Mortgagor's obligation under the Mortgage secured by the Condominium.
- 10.8 <u>Director and Officer Liability Insurance.</u> To the extent such insurance is available at reasonable cost, the Association shall purchase and maintain insurance in an amount of no less than One Million Current Dollars (\$1,000,000.00) on behalf of any director, officer, or member of a committee of the Association (collectively, the "agents") against any liability asserted against or incurred by the agent in such capacity or arising out of the agent's status as such, regardless of whether the Association would have the power to indemnify the agent against such liability under applicable law.

- 10.9 <u>Fidelity</u>. A fidelity bond for Directors, officers, trustees, and employees responsible for handling funds collected and held for the Association or for the Owners, naming as an insured the Association for an amount sufficient to cover at least twenty-five percent (25%) of the Association's estimated annual operating expenses and reserves.
- **10.10** <u>Limitation of Liability for Failure to Obtain Insurance.</u> Neither the Association, the Board, Declarant, nor any managing agent shall be liable for failure to obtain any coverages required by this Article 10, or for any loss or damage resulting from such failure, if such failure is due to the unavailability of such coverages from reputable insurance companies, or if such coverages, in the judgment of the Board, are so available only at unreasonable cost.
- **10.11** <u>Notice to Owners.</u> The Board of Directors shall promptly furnish to each Owner written notice of the procurement of, subsequent changes in, or termination of, insurance coverages obtained on behalf of the Association, or of any decision not to procure any insurance coverages described in this Article 10.

10.12 Generally Applicable Requirements.

- (a) Each insurance policy or fidelity bond obtained by the Association shall provide that:
- (i) The insurer waives any right to claim by way of subrogation against the Ground Lessor, Declarant, the Association, the Board of Directors, any managing agent, or the Owners and their respective invitees, agents, employees, tenants, and, in the case of Owners, members of their households.
- (ii) Such policy shall not be cancelled, invalidated, or suspended as a result of the conduct of any Owner (including his or her invitees, agents, employees and tenants) or any member of the Board of Directors, or any managing agent, or any officer or employee of the Association, unless a prior demand in writing that the Board of Directors or the managing agent cure the defect has been given and neither the Board of Directors nor the managing agent has cured such defect within sixty (60) calendar days after such demand.
- (iii) Such policy may not be cancelled or substantially modified (including cancellation for nonpayment of premium) without at least thirty (30) calendar days' prior written notice to the Association and any managing agent and, in the case of liability and physical damage insurance and fidelity bonds or insurance, to all First Mortgagees.

- (b) Declarant shall be covered by each policy obtained by the Association, provided that such coverage is customarily available at reasonable cost.
- (c) All policies of insurance shall be written by reputable companies licensed to do business in the State of California.
- (d) The deductible, if any, on any insurance policy purchased by the Association shall be a common expense, provided that the Association may assess as a special assessment against an Owner any deductible amount necessitated by the negligence, misuse, or neglect of such Owner.
- 10.13 <u>Umbrella Liability Insurance</u>. The Association shall obtain and maintain a policy of umbrella liability insurance to supplement all other insurance policies maintained by the Association. The limits of such insurance policy shall not be less than that amount that is required to ensure that the total liability insurance carried by the Association will equal Five Million Current Dollars (\$5,000,000.00).
- **10.14** <u>Increases in Coverage.</u> Subject to the provisions of Section 9.6(b) of this Declaration, the Board may, when in its best judgment it deems it advisable, increase the amount of any coverages carried pursuant to this Article 10.

ARTICLE 11

DESTRUCTION OF IMPROVEMENTS

- 11.1 <u>Minor Repair and Reconstruction.</u> The Association shall have the duty to repair and reconstruct any structure or other improvement in the Project, without the consent of Owners and irrespective of the amount of available insurance proceeds, in all cases of partial destruction when the estimated cost of repair and reconstruction does not exceed One Hundred Thousand Current Dollars (\$100,000.00).
- of Reconstruction Costs. If (i) there is a total or partial destruction of any structure or other improvement in the Project, (ii) the estimated cost of repair and reconstruction of such improvements exceeds One Hundred Thousand Current Dollars (\$100,000.00), and (iii) the available proceeds of the insurance carried pursuant to Article 10 of this Declaration are sufficient to cover at least eighty-five percent (85%) of the estimated cost of repair and reconstruction, the Association shall promptly rebuild such improvement. The Association shall execute, acknowledge, and record in the Official Records of the County Recorder of Santa Cruz County, California, not later than one hundred twenty

(120) calendar days from the date of destruction, a certificate declaring the intention of the Association on behalf of the Owners to rebuild.

- Destruction When Insurance Proceeds Are Less Than Eighty-Five Percent (85%) of Reconstruction Costs. If (i) there is a total or partial destruction of any of the structures or other improvements in the Project, (ii) the estimated cost of repair and reconstruction of such improvements exceeds One Hundred Thousand Current Dollars (\$100,000.00), and (iii) the available proceeds of the insurance carried pursuant to Article 10 of this Declaration are less than eighty-five percent (85%) of the estimated cost of repair and reconstruction, the Association shall promptly rebuild such improvements unless, within ninety (90) calendar days from the date of destruction, Owners then holding at least seventy-five percent (75%) of the total voting power of Owners (excluding Declarant during the Declarant Control Period) present and entitled to vote, in person or by proxy, at a duly constituted meeting, determine that repair and reconstruction shall not take place, and First Mortgagees of Condominiums that have at least sixty-seven percent (67%) of the votes of all Condominiums encumbered by First Mortgages of which the Association has notice and Declarant, during the Declarant Control Period, give their written consent that repairs and reconstruction shall not take place. If such a meeting of Owners is called, the Association shall solicit and obtain bids from at least two reputable contractors to repair and reconstruct the improvements in accordance with the original plans and shall present this information to the Owners at the meeting. If repair and reconstruction are to take place, the Association shall execute, acknowledge, and record in the Official Records of the County Recorder of Santa Cruz County, California, not later than one hundred twenty (120) calendar days from the date of destruction a certificate declaring the intention of the Owners to rebuild.
 - 11.4 Repair and Reconstruction of Improvements.
 - (a) In any case in which improvements are to be repaired or rebuilt:
- (i) Each Owner shall be obligated to contribute his or her Percentage Interest of the cost of repair and reconstruction over and above available insurance proceeds. If any Owner fails to pay his or her proportionate share, the Board may enforce the assessment pursuant to the provisions of Sections 9.12 through 9.14 of this Declaration.
- (ii) The Association shall, after having obtained bids from at least two reputable contractors, award the repair and reconstruction work to the bidder that best meets the requirements for the performance of such work as set forth by the

Association. The Association shall have the authority to enter into a written contract with the contractor for the repair and reconstruction, and the insurance proceeds held by the Association or a trustee, as applicable, shall be disbursed to the contractor according to the terms of the contract.

- (iii) Each Owner shall be further obligated to execute any amendments to the Condominium Plan necessary to conform such Plans to the Project as then rebuilt. If any Owner fails to execute the necessary documents, Declarant, during the Declarant Control Period, and thereafter the Association, or individuals authorized by the Board, shall act as attorney-in-fact of all Owners to execute such documents on his or her behalf.
- (b) In any case in which improvements are to be repaired or rebuilt, such repaired or rebuilt improvements will be reasonably compatible in terms of quality, materials and architectural style with the improvements, including Units and Common Areas, remaining in the Project.

11.5 Purchase After Election Not To Rebuild; Restoration of Site.

- (a) If (i) the estimated cost of repair and reconstruction exceeds One Million Current Dollars (\$1,000,000.00), (ii) the available insurance proceeds are less than eighty-five percent (85%) of such estimated cost, and (iii) the Owners, with the consent of the applicable number of First Mortgagees and Declarant, during the Declarant Control Period, elect not to rebuild, the Association shall have the right, after obtaining the consent, by affirmative vote of seventy-five percent (75%) of the total voting power of Owners present and entitled to vote, in person or by proxy, at a duly constituted meeting, to purchase the Condominiums of which the Units were damaged or destroyed.
- (b) The price for any such Condominium which shall be paid by the Association to its Owner shall be the lesser of (i) the replacement cost of the Condominium, as determined by the Board based on bids to repair or restore the Condominium received from at least two (2) reputable contractors, less the insurance proceeds payable to the Owner, or (ii) the purchase price as established pursuant to Section 14.2(b) to be established as of a date ninety (90) calendar days after the date of the damage or destruction.
- (c) Each Owner, other than those whose Condominiums are purchased, shall be obligated to contribute his or her proportionate share, based on the ratio of the square footage of the floor area of the Unit to be assessed to the total square footage of

floor area of all Units in the Project other than the Units to be purchased, of the cost, over and above available insurance proceeds, to purchase the Condominiums of which the Units were damaged or destroyed and to restore the site on which such Condominiums were located pursuant to Section 11.5(g). If any Owner fails to pay his or her proportionate share, the Board may enforce the assessment pursuant to the provisions of Sections 9.12 through 9.14 of this Declaration.

- (d) Any payment of the purchase price to an Owner whose Condominium is purchased by the Association shall be made jointly to such Owner and all Mortgagees of such Owner's Condominium.
- (e) By accepting a deed to a Condominium, each Owner agrees to be bound by the provisions of this Section 11.5 and to sell his or her Condominium and to convey it by grant deed to the Association as provided in this Section 11.5.
- (f) Concurrently with any such purchase, each Owner agrees to execute any documents required to effectuate such purchase, including any amendments to the Condominium Plan and any appropriate supplemental Condominium Plan necessary to eliminate from the Project the purchased Condominiums as Common Areas. If any Owner fails to execute the necessary documents, Declarant, during the Declarant Control Period, and thereafter the Association, or individuals authorized by the Board, shall act as attorney-in-fact of all Owners to execute such documents on such Owner's behalf.
- (g) The Association shall cause to be removed, at its expense, any debris or rubble remaining after the total of partial destruction of any Condominiums that it purchases and shall restore the site upon which such Condominiums were located to a condition compatible with the continuation of the rest of the Project.
- 11.6 <u>Application of Insurance Proceeds.</u> Except in any case in which improvements are not to be repaired or reconstructed pursuant to this Article 11, all insurance proceeds, including those payable to Mortgagees, shall be applied to the repair and reconstruction of damaged improvements. If improvements are not to be repaired or replaced, the insurance proceeds shall be apportioned among the Owners and their respective Mortgagees in proportion to the relative value of the Units, as determined pursuant to the procedure set forth in Section 12.1, below.
- 11.7 <u>Termination of Partition Waiver.</u> In the event of damage or destruction which (a) renders more than fifty percent (50%) of the Units uninhabitable, or (b) results in an estimated cost of repair and reconstruction that exceeds One Million Current

Dollars (\$1,000,000.00), and if the Owners decide not to repair or rebuild, either by calling a meeting and rejecting all bids and estimates presented, or by failing to call a meeting, or by failing to commence the work of repair or rebuilding within twelve (12) months or to complete the work of repair or rebuilding within twenty-four (24) months from the date of damage or destruction, the prohibition against judicial partition of the Project pursuant to California Civil Code section 1359 shall terminate, and the Board shall, or if it does not, any Owner, or the Mortgagee of any Owner may, record a sworn declaration setting forth such decision and stating that the prohibition against judicial partition set forth in this Declaration has terminated. On recordation of such declaration, the right of any Owner to bring an action for judicial partition shall revive immediately.

11.8 <u>Distribution of Excess Insurance Proceeds.</u> Any amount in excess of the insurance proceeds necessary to repair and reconstruct damaged improvements pursuant to Section 11.2 of this Declaration or to purchase Condominiums pursuant to Section 11.5 of this Declaration shall be obtained by the Association for distribution to all of the Owners who were Members of the Association at the time that the improvements were totally or partially destroyed. Each qualifying Owner's proportionate share of any excess proceeds shall be determined by applying the Percentage Interest of each Owner to such proceeds or to reduction of assessments.

ARTICLE 12

CONDEMNATION; PARTITION

12.1 <u>Condemnation.</u> Any taking of all or part of the Project by eminent domain shall be governed by Section 17 of the Common Areas Lease, as that section may from time to time be amended. In the event of such a taking, all compensation and damages relating to the Project (as distinct from the taking of an individual unit) shall be payable to the Association as trustee for all affected Condominium Owners according to the relative value of their respective interests, as they may appear. For purposes of this Section 12.1, "relative value" shall be calculated using the prices for the affected Condominiums determined pursuant to section 14.2(b), below. The Association shall give prompt written notice of condemnation proceedings to any Mortgagee of a Condominium who has filed a request for notices pursuant to Section 15.2 herein. Any compensation and damages payable to an Owner of a Condominium shall be subject to (i) the rights of each Mortgagee holding a Mortgage encumbering such Unit to be paid from the compensation and damages apportioned to such Unit, as their interests may appear, and (ii) all unpaid regular and/or special assessments together with any interest charges attributable thereto.

12.2 <u>Waiver of Right to Partition.</u> To the maximum extent permitted by law, an Owner shall have no right to partition or divide his or her ownership of the Common Areas, provided that any partition of a co-tenancy in a Condominium shall be by sale or appraisal and not by partition in kind.

ARTICLE 13

SPECIAL RESTRICTIONS UPON POSSESSION, OCCUPANCY AND USE OF UNITS

- 13.1 <u>Purpose.</u> The intent of this Article 13 is to create and perpetuate a stock of affordable housing available to the faculty of the University of California at Santa Cruz ("UCSC"), in order to support and enhance the educational community at UCSC. The provisions of this Article 13 shall apply to and bind all Condominiums in the Project and shall be construed liberally to effectuate such purpose.
 - **13.2** [RESERVED].
 - **13.3** [RESERVED].
- 13.4 Qualified Use. No person other than Declarant shall possess, occupy or use any Condominium unless such person is a "Qualified Person" a "Conditionally Qualified Person" as defined in this Article 13. A person shall be deemed to possess, occupy or use a Condominium for purposes of this Article if he or she resides in a Condominium or otherwise regularly or from time to time physically occupies or uses it.
- 13.5 Qualified Person. A "Qualified Person" is a person determined to be eligible either individually or as a member of a class by the Chancellor of UCSC in accordance with the employee housing policy and criteria of UCSC, as amended from time to time. That policy and those criteria are intended and dedicated to further the educational purposes of UCSC by enhancing the quality of education through retention and recruitment of faculty and staff members (as defined by the Chancellor of UCSC); therefore, the Chancellor of UCSC may, in his or her sole discretion, from time to time designate as Qualified Persons, those employees of Declarant who are an integral part of the UCSC community, and with respect to whom the Chancellor of UCSC shall have determined that providing adequate housing is a priority.

Following designation as a Qualified Person, a person shall continue to be a Qualified Person until his or her employment by The Regents terminates, at which time he or she will no longer be a Qualified Person; provided, however, that no Owner who

is a Qualified Person (a "Qualified Owner") shall lose his or her status as a Qualified Person because of a termination of employment as a result of disability or retirement. "Disability" includes, but is not limited to, the following: (i) A physical or mental impairment that substantially limits one or more of a person's major life activities. (ii) A record of having, or being perceived as having, a physical or mental impairment, but not including current illegal use of, or addiction to, a controlled substance (as defined by section 102 of the federal Code of Controlled Substance Act, 21 U.S.C. § 802). "Retirement" shall mean termination of employment at a time when a person shall be eligible to receive retirement benefits from the University of California Retirement System, or any other retirement system used by Declarant.

- 13.6 Conditionally Qualified Persons. The following persons ("Conditionally Qualified Persons") shall be considered to be Qualified Persons only upon the conditions and for the periods of time specified in this Section:
- (a) Any Qualified Owner shall continue to be qualified until the expiration of six (6) months following the date on which such person shall cease to have been a Qualified Person as defined in Section 13.5, above;
- (b) The spouse or Domestic Partner of a Qualified Owner who shall possess, occupy and use a Unit as of the date of death of such Qualified Owner shall continue to be a Qualified Person for as long as he or she maintains the Unit as his or her principal place of residence or until remarriage (or entry into a domestic partnership), if earlier;
- (c) The dependent child of a Qualified Owner who shall possess, occupy and use a Unit as of the date of death of such Qualified Owner shall continue to be a Qualified Person for three (3) years from such date or, if earlier, until the then current Owner sells or disposes of the Unit or, in the case of a minor dependent child, until such child attains the age of twenty-one (21) years;
- (d) An heir or legatee (other than a spouse, Domestic Partner, or dependent child) of a Qualified Owner who shall acquire ownership of a Condominium from the estate of such Qualified Owner shall be a Qualified Person for one (1) year from the date such heir or legatee shall acquire record ownership of such Condominium;
- (e) A Qualified Owner's spouse who acquires the sole rights of occupancy of a Unit pursuant to any marital settlement, proceeding or decree shall be a Qualified Person for a period of one (1) year from the date of such settlement, proceeding or decree;

- (f) A co-Owner of a fifty percent (50%) or less undivided interest in a Condominium who resides in such Condominium during the lifetime of a Qualified Owner shall be a Qualified Person for a period of one (1) year from the death of such Qualified Person;
- (g) Any lessee of a Condominium pursuant to Section 14.7 of this Declaration during the term of a lease thereunder; and
- (h) Any purchaser of a Condominium who purchased such Condominium on the open market pursuant to Section 14.1(e), or who purchased it at a sale by or on behalf of a Mortgagee pursuant to Section 14.4(a) for so long as such purchaser shall occupy such Condominium as his or her principal residence.

A Conditionally Qualified Person, as defined in this Section, may continue to possess, occupy and use a Condominium until the expiration of his or her Conditionally Qualified status pursuant to this Section. No spouse, heir or legatee of a Conditionally Qualified Person shall also become a Conditionally Qualified Person by reason of the death or marital dissolution of such Conditionally Qualified Person.

13.7 Offer or Sale by Conditionally Qualified Person.

- (a) Any Conditionally Qualified Person who owns a Condominium may, at any time during the period of conditional qualification, either:
- (i) Offer to sell such Condominium to Declarant at the price and on the terms set forth in Section 14.2 hereof; or
- (ii) Sell or transfer his or her entire ownership interest in the Condominium to an Approved Purchaser designated pursuant to Section 14.2, to be eligible to purchase such Unit at the price and on the terms set forth in Section 14.2 hereof.
- 13.8 Qualified Users. "Qualified User" means a person who owns a Condominium, and is either a Qualified Person or a Conditionally Qualified Person. Use of a Condominium by a person who is (a) related by marriage, blood, adoption, domestic partnership, guardianship or foster care to a Qualified User, (b) co-owner of a fifty percent (50%) or less undivided interest in a Condominium, the remainder of which is owned by a Qualified User, (c) an invitee or guest of a Qualified User, or (d) a domestic or nurse employed by a Qualified User, during any period during which such Qualified User uses the Condominium as his or her principal place of abode shall be

deemed to be a use thereof by the Qualified User. Use of a Condominium by a person pursuant to a lease approved by Declarant between such person and a Qualified Person shall also be deemed to be use thereof by the Qualified Person.

13.9 <u>Enforcement of Qualified Use Restriction.</u> Declarant, as well as any Owner or the Association, may enforce the provisions of this Article by appropriate action, including, but not limited to, any remedy or remedies provided in Section 4.28 of this Declaration. All the rights and remedies available to the Association under Article 7 with respect to any violation of this Article are also available to Declarant.

ARTICLE 14

LIMITATIONS ON TRANSFER OF CONDOMINIUMS

- **14.1** <u>Right of First Refusal.</u> Declarant shall have a right of first refusal with respect to the sale or other transfer of any Condominium by an Owner other than Declarant, as follows:
- (a) If at any time an Owner shall intend to sell or assign his or her Condominium or any interest therein in excess of a fifty percent (50%) undivided interest, or to reduce his or her ownership interest therein to less than a fifty percent (50%) undivided interest, such Owner shall first offer to sell said Condominium to Declarant at the price and on the terms set forth in Section 14.2; provided, however, that an offer by an Owner to sell or assign his or her Condominium to an Approved Purchaser, as that term is defined in Section 14.5, and an acceptance thereof by the offeree, at the price and on the terms and conditions set forth in Section 14.2, shall be deemed to satisfy all obligations of such Owner pursuant to Section 14.1(a).
- (b) Each offer to sell a Condominium pursuant to this Section 14.1 shall be made in a written notice delivered to Declarant.
- (c) Declarant shall have the right, but not the obligation, to assign its right of acceptance of any offer made pursuant to this Section 14.1 to an Approved Purchaser, as hereinafter defined.
- (d) If Declarant rejects said offer or does not respond within forty-five (45) calendar days, an Owner wishing to sell or transfer an interest in the Condominium may, by appropriate notices or advertisement, documentation of which shall be promptly delivered to Declarant, offer the Condominium for sale for a period of sixty (60) calendar days, at the price and on the terms and conditions set forth in Section 14.2,

to all of UCSC's faculty, academic staff, and salaried employees. An offer made pursuant to this Section shall remain open until accepted or, in the absence of an acceptance, for sixty (60) calendar days from the date of publication of the original notice or advertisement. All offers made under this Section are subject to the requirement that only a Qualified Person (as defined in Section 13.5) who has been designated to be an "Approved Purchaser" pursuant to Section 14.5 may accept such an offer.

- (e) If an Owner's offer pursuant to Section 14.1(d), above, is not accepted within sixty (60) calendar days after it is made, and if the Owner still wishes to sell or transfer an interest in the Condominium, the Owner shall give notice to Declarant of his or her intention and provide Declarant with the opportunity to agree to purchase the Condominium for a period of fifteen (15) calendar days at a price and on terms which are not more favorable to the Owner than Owner's original offer made pursuant to Section 14.1(a), above. After the expiration of the fifteen (15) calendar day period, the Owner may offer the Condominium for sale on the open market at any price, and on any terms. Such an offer by an Owner following expiration of not fewer than one hundred twenty (120) calendar days from the receipt of the initial notice to Declarant given by Owner pursuant to this Section may be made to any person, without respect to their "qualification" under Article 13 or their designation as an Approved Purchaser under Section 14.5. Any such person who purchases the Condominium shall, however, be bound by all the provisions of this Article 14 respecting any subsequent resale of the Condominium.
- (f) Any written notice of acceptance given by Declarant or an assignee of Declarant to an Owner who shall have made an offer under this Section 14.1 shall constitute a binding acceptance of such offer at the price and on the terms set forth in Section 14.2 hereof.
- (g) Any purported offer or sale of a Condominium other than in conformance with this Section 14.1, shall be void and of no effect, except that such purported offer, sale or other transfer shall constitute an irrevocable offer to Declarant to sell Owner's Condominium at the price and on the terms set forth in Section 14.2, below, which offer shall be deemed to be delivered to Declarant thirty (30) calendar days following the date Declarant receive actual notice of such purported offer, sale or transfer.
- **14.2** <u>Purchase Price and Terms.</u> The purchase price of any Condominium offered or sold by a person other than Declarant pursuant to this Article 14 shall be the lesser of:

- (a) The fair market value of the Condominium as mutually determined by the Owner and Declarant, either by agreement or, in the absence of agreement, pursuant to the definition of "fair market value" below; or ; or
- (b) The sum of: (i) the purchase price of the Condominium paid by the Owner, plus (ii) the product of the purchase price of the Condominium paid by Owner and the larger of either (x) the fractional change in the Consumer Price Index (as defined below), or (y) the Faculty Salary Index which is that index published and maintained by Declarant's Office of Loan Programs, each as issued or in effect for the month immediately preceding the date on which the Owner purchased the Condominium and said Index or level as issued or in effect for the month preceding the date of the offer made pursuant to Section 14.1, plus (iii) the cost of all capital improvements less depreciation to the Condominium made by the Owner, but only to the extent that each such capital improvement exceeded \$1,000 in cost in Current Dollars including such costs as may be distributed to an Owner's unit as a result of qualified capital improvements made by the Homeowner's Association and to the extent that the value of such alteration and its reasonable life for depreciation purposes was certified by the ARB prior to completion of such improvement following submission by the Owner of cost documentation in such form as the ARB may prescribe, which documentation shall be subject to audit and proof, less (iv) the reasonable cost to cure any failure to maintain the Unit or waste or destruction thereof in excess of normal wear and tear. As used in this Section, "Consumer Price Index" means the National Consumer Price Index for All Urban Consumers, All Items (1982-84=100) ("CPI-U") as published by the United States Department of Labor, Bureau of Labor Statistics, or, if such index ceases to be published, then any comparable successor index as determined by Declarant which measures changes in the prices of consumer items.

"Fair market value" for purposes of Section 14.2(a) above, may, for purposes of this Section 14.2 be determined by an appraisal conducted by an appraiser mutually satisfactory to Owner and Declarant, or in the absence of a mutually satisfactory appraiser, the fair market value shall be the average of appraisals conducted by two (2) appraisers each of which shall meet the requirements set forth Section 12.1 (i)-(iii)above, one selected by the Owner and one selected by Declarant. All appraisers shall consider the effect of the resale restrictions of this Declaration, including Section 14.2 above and the duration of the term of the Common Areas Lease.

No offer subject to this Article 14, other than an open market offer pursuant to Section 14.1(e) shall require an earnest money deposit of more than two percent (2%) of

the purchase price nor require the closing of escrow for the sale of a Unit or final payment therefor in fewer than forty-five (45) calendar days from the date of acceptance by the purchaser.

All sales of Condominiums by Declarant shall be exempt from this Section 14.2.

- 14.3 <u>Closing Payment.</u> If any offer made pursuant to Section 14.1 of this Article is accepted by Declarant or an Approved Purchaser, as defined in Section 14.5, the closing thereon shall occur within ninety (90) calendar days of acceptance of Owner's offer, and the purchase price shall be paid in cash as follows:
- (a) To the Association, for due and unpaid assessments, if any, to the date of closing;
- (b) To Owners' First and subsequent Mortgagees, if any, as their respective interests may appear, or by assumption of the First or Second Mortgages as arranged by the purchaser;
- (c) In payment of necessary closing costs customarily charged to sellers, including by way of example, but not limitation, broker fees, title insurance premiums, documentary transfer tax and escrow fees; and
 - (d) The remainder to Owner.

Real property taxes and assessments are to be prorated to the date of closing. The purchase, possession, occupancy and use of the Condominium shall be subject to all the terms and conditions of this Declaration.

14.4 Secured Parties.

(a) Except as provided in this Section 14.4, this Article shall not apply to a transfer to or a purchase by a Mortgagee or an Institutional Mortgagee which acquires its title as a result of holding a Mortgage upon the Condominium concerned, and this exception shall be effective whether the title is acquired (i) by deed from the Owner or his or her successor in title, or (ii) through foreclosure proceedings. However, any Condominium Mortgagee or other acquiree of a Condominium of an Owner pursuant to foreclosure, assignment in lieu of foreclosure or other, substantially similar proceedings may assign or transfer the same only in accordance with the terms and conditions of Article 14 of the Common Areas Lease.

- (b) Except as provided in this Section 14.4, this Article 14 shall not apply to an acquisition of a Condominium by a purchaser, other than those described in Section 14.4(a) above, who acquires title to a Condominium at a duly advertised public sale with open bidding which is provided by law, such as, but not limited to, execution sales, foreclosure sales, judicial sales or tax sales, nor to the heirs, successors or assigns of such a purchaser.
- 14.5 Approved Purchasers. Declarant, or by designation of Declarant, the Chancellor of UCSC, may compile a list of Qualified Persons who shall be determined by Declarant or by the Chancellor of UCSC to be eligible, in descending order of priority, as listed, to purchase a Condominium from an Owner at a price not to exceed the price calculated for such Condominium pursuant to Section 14.2 above (such persons to be "Approved Purchasers"). No person shall be deemed to be an Approved Purchaser with respect to any offer, sale or transfer of a Condominium unless an offer to sell such Condominium has, in good faith, been made to each Approved Purchaser having a higher priority as determined by Declarant or the Chancellor of UCSC.
- 14.6 <u>Certification of Status.</u> Upon written request of any prospective transferee or purchaser, tenant or an existing or prospective Mortgagee, transferee or beneficiary of any Condominium, Declarant shall, within fifteen (15) calendar days thereof, issue a written and acknowledged certificate in recordable form, evidencing if such be the case:
- (a) With respect to a proposed sale under this Article 14, that the proper offers were made by the selling Owner and that Declarant did not elect to exercise its right to purchase;
- (b) With respect to a sale to an Approved Purchaser, that proper notice was given by the Owner and that such sale was permissible hereunder; or
- (c) That a deed-in-lieu of foreclosure was, in fact, given in lieu of foreclosure and is not subject to this Article 14, except as provided in Section 14.4 hereof.
- 14.7 <u>Leasing.</u> An Owner other than Declarant may lease or rent a Condominium only on the following terms and conditions:
- (a) No Condominium shall be leased or rented except in accordance with this Section 14.7.
- (b) Owners may freely enter into any lease or rental agreement of a Condominium which does not have a term longer than twelve (12) months or which,

when added to the term of any prior lease or rental agreement, would result in occupancy by tenants of more than thirteen (13) months within the preceding thirty-six (36) months without prior written consent of Declarant; provided, however, that if such lease or rental agreement shall be entered in conjunction with an academic leave of Owner approved by the Chancellor of UCSC, the occupancy of the Condominium by tenants may extend for the duration of such leave.

- (c) Execution by an Owner of any lease or rental agreement of a Condominium in violation of Section 14.7(b), above, without the prior written consent of Declarant shall be deemed to be an irrevocable offer to sell the Condominium to Declarant at the price and on the terms set forth in Sections 14.2 and 14.3.
- 14.8 <u>Unauthorized Sales or Transfers.</u> If any Owner shall attempt to sell, assign or otherwise transfer the Owner's Condominium to any person other than an eligible Qualified Person and Approved Purchaser without making the offers described in Section 14.1 hereof or without otherwise following the procedures set forth in this Article 14, such attempted or purported sale, assignment or transfer shall be wholly null and void and shall confer no title or interest whatsoever upon the purported purchaser or transferee. If any Owner shall attempt to transfer an interest in his or her Condominium to a person other than a Qualified Person and Approved Purchaser and such transfer shall (a) reduce the undivided ownership interest in the Condominium of the Owner to less than fifty percent (50%), or (b) deprive the Owner of the right of immediate occupancy at any time in the future, such transfer of estate and right to possession shall be wholly null and void and shall confer no title, interest or right of possession or occupancy whatsoever upon the purported transferee in the absence of consent thereto by Declarant.
- 14.9 <u>Termination of Articles 13 and 14.</u> Articles 13 and 14 of this Declaration shall cease to have any effect or confer any power on any person with respect to a particular Unit on or after the first to occur of (a) the termination of this Declaration and any renewal thereof or (b) the election of Declarant, exercisable at any time after December 31, 2063,, evidenced by recordation of a written notice of termination of Articles 13 and 14, or (c) termination of Declarant's status as Ground Lessor.
- **14.10** Enforcement of Article 14. Declarant, as well as any Owner and the Association, may enforce the provisions of this Article by appropriate action, including, without limitation, an action for unlawful detainer or to enjoin trespass. All the rights and remedies available to the Association under Article 13 with respect to any violation of this Article 14 shall also be available to Declarant.

14.11 Effect of Article 14. Nothing in this Article 14, and no act or failure to act of any person under the provisions of this Article, shall affect or work to limit, suspend or abridge the provisions of Article 13.

ARTICLE 15

PROTECTION OF MORTGAGEES

- **15.1** <u>Mortgage Permitted.</u> Any Owner may encumber his or her Condominium with a Mortgage.
- 15.2 <u>Notice to Board of Directors.</u> An Owner who mortgages his or her Condominium shall promptly notify the Board of Directors of the name and address of his or her Mortgagee. In addition, any Mortgagee desiring to receive notices pursuant to Section 15.9 of this Declaration shall file a written request for such notices with the Board of Directors, stating both its name and address and the Identifying Number or address of the Condominium on which it holds a Mortgage.
- <u>Subordination</u>. Any lien created or claimed under the provisions of this Declaration is expressly made subject and subordinate to the rights of any First Mortgage that encumbers all or a portion of the Project, or any Condominium, made in good faith and for value, provided that such lien was not evidenced by a notice of delinquent assessment recorded prior to the recordation of such First Mortgage. On foreclosure of such Mortgage, such lien for assessments or the installments that have accrued up to the time of foreclosure shall be subordinate to the lien of such Mortgage, with the foreclosure-purchaser taking title to the Condominium free of such lien for assessments or installments that have accrued up to the time of the foreclosure sale. On taking title to the Condominium, the foreclosure-purchaser shall be obligated to pay only assessments or other charges levied or assessed by the Association that become due or payable on or after the date the foreclosure-purchaser acquired title to the Condominium, provided that, if all Owners are required to pay their proportionate share, the assessments or charges that become due or payable on or after the date the foreclosure-purchaser acquired title to the Condominium may include amounts attributable to previously due but unpaid assessments.
- 15.4 <u>Control of Amendment of Condominium Instruments.</u> Unless a greater percentage is expressly required by this Declaration, the Articles, or the Bylaws, or by law, the prior written consent (or deemed consent as provided below in this clause) of sixty-seven percent (67%) of the voting power of the Association as defined in Section 8.2 of this Declaration and of First Mortgagees of Condominiums that have at least fifty-

one percent (51%) of the votes of all Condominiums encumbered by First Mortgages of which the Association has notice shall be required to add or amend any material provisions of the Condominium Instruments, which establish, provide for, govern, or regulate any of the following:

- (a) Voting rights.
- (b) Increase in assessment of more than twenty-five percent (25%), collection of assessments, assessment liens, or the priority or subordination of such liens.
- (c) Relocation of reserves for maintenance, repair, and replacement of Common Areas or improvements located thereon.
 - (d) Casualty and liability insurance or fidelity bonds.
 - (e) Reallocation of, or changes in, the rights to use the Common Areas.
- (f) Responsibility for maintenance and repair of Condominiums and Common Areas and the improvements thereon.
- (g) Expansion or contraction of the Project, or the addition, annexation, or withdrawal of real property to or from the Project.
 - (h) Boundaries of any Unit.
- (i) The interests or rights of the Association or Owners in and to the Common Areas.
- (j) The convertibility of Units into Common Areas or of Common Areas into Units.
 - (k) The leasing of Units.
- (l) Imposition of any additional right of first refusal or similar restriction on the right of an Owner to sell, transfer, or otherwise convey his or her Condominium.
 - (m) Destruction and restoration of improvements.
- (n) Termination of the Project following substantial destruction or condemnation of improvements.

- (o) Hazard or fidelity insurance requirements.
- (p) Any provisions that are for the express benefit of First Mortgagees or insurers or governmental guarantors of First Mortgages.

For purposes of this provision, an addition or amendment shall not be considered material if it is for the purpose of correcting technical errors, or for clarification only. Any First Mortgagee who receives a written request to consent to additions or amendments requiring consent under this provision who does not deliver or post to the requesting party a negative response within thirty (30) calendar days after such receipt shall be deemed to have consented to such request.

- 15.5 Restrictions on Certain Changes. In addition to all other approvals and consents required to be obtained from Mortgagees under this Declaration, unless at least sixty-seven percent (67%) of First Mortgagees of Condominiums encumbered by First Mortgages of which the Association has notice, and Declarant during the Declarant Control Period have given their prior written approval, neither the Association nor the Owners shall be entitled:
- (a) By act or omission to seek to abandon or terminate the Project, except for abandonment provided by statute in case of condemnation of or substantial loss to the Units and Common Areas.
- (b) To change the pro rata interest or obligations of any Condominium in order to levy assessments or charges, allocate distribution of hazard insurance proceeds or condemnation awards, or determine the pro rata share of ownership of each Condominium in the Common Areas, except with regard to the expansion of the Project as provided for by this Declaration.
- (c) To partition or subdivide any Unit after a deed conveying title to the Condominium containing such Unit has been recorded.
- (d) By act or omission to seek to abandon, partition, subdivide, encumber, sell, or transfer the Common Areas. The granting of easements, licenses, permits, or rights-of-way for public utilities or for other public or quasi-public purposes consistent with the intended use of the Common Areas by the Association, Declarant, or the Owners shall not be deemed to be a transfer within the meaning of this clause. Further, this subsection 15.5(d) shall not apply to the extent necessary to permit expansion of the Project as provided for by this Declaration.

- (e) To use hazard insurance proceeds for losses to Units or Common Areas or to any other Association property, for other than the repair, replacement, or reconstruction of such improvements or property, except as provided by statute or this Declaration in case of substantial loss to the Units or Common Areas of the Project.
- all insurers or guarantors of First Mortgages shall have the right to examine the Condominium Instruments and the books and records of the Association and the right to require the submission of financial data concerning the Association, including annual audit reports, budgets, and operating statements, as furnished to the Owners. In addition, if an audited financial statement for an annual accounting period is not available for the Association, any First Mortgagee shall be allowed to have an audited statement prepared at its own expense.
- 15.7 Priority in Distribution of Proceeds. Subject to the obligations of Mortgagees regarding rebuilding as set forth in Section 11.6 of this Declaration, no Owner, or any other party, shall have priority over any right of Institutional First Mortgagees of Condominiums pursuant to their Mortgages in case of a distribution to Owners of insurance proceeds or condemnation awards for losses to or a taking of Units or Common Areas. Any provision to the contrary in the Condominium Instruments is to such extent void. All applicable fire and all physical loss or extended coverage insurance policies shall contain loss payable clauses acceptable to the affected Institutional First Mortgagees, naming the Mortgagees as their interests may appear.
- 15.8 <u>Status of Improvements.</u> All improvements in the Common Areas (such as recreation and service areas) shall be available for use by Owners, and all such improvements with respect to which regular or special assessments for maintenance or other uses may be levied shall constitute Common Areas. All such improvements shall be owned by the Association free of encumbrances except for the Common Areas Lease and any easements, licenses, permits, or rights-of-way granted for utilities or for other public or quasi-public purposes consistent with the intended use of such property by the Owners or by the Association.

15.9 Notices to Mortgagees.

(a) Subject to the provisions of Section 9.13 of this Declaration, if any Owner is in default under any provision of this Declaration or under any provision of the Articles, the Bylaws, or the Association Rules and Regulations, and the default is not cured within sixty (60) calendar days after written notice to that Owner, the Association shall give to any Mortgagee who has filed a request for notices with the Board of

Directors pursuant to Section 15.2 of this Declaration, notice of such default and of the fact that the sixty (60) calendar day period has expired.

- (b) Any Mortgagee who has filed a request for notices with the Board of Directors pursuant to Section 15.2 of this Declaration also shall be sent a written notice by the Board of Directors of:
- (i) Any condemnation or casualty loss that affects either a material portion of the Project or the Condominium on which such Mortgagee holds a Mortgage.
- (ii) Any lapse, cancellation, or material modification of any insurance policy or fidelity bond maintained by the Association.
- (iii) Any proposed action that requires the consent of a specified percentage of any class of Mortgagees which includes the Mortgagee requesting notices.
- **15.10** <u>Lien Not Invalidated.</u> No breach of any provision of this Declaration shall invalidate the lien of any Mortgage made in good faith and for value; but all of the covenants, conditions, and restrictions shall be binding on any Owner whose title is derived through foreclosure sale, trustee's sale, or otherwise.
- 15.11 <u>Status of Loan to Facilitate Resale.</u> Any First Mortgage given to secure a loan to facilitate the resale of a Condominium after acquisition by foreclosure or by a deed-in-lieu of foreclosure or by an assignment in lieu of foreclosure shall be deemed to be a loan made in good faith and for value and the holder of such loan shall be entitled to all of the rights and protection of Mortgagees under this Declaration.
- **15.12** Right to Appear at Meetings. Because of its financial interest in the Project, any Mortgagee may appear (but cannot vote) at meetings of Owners and the Board to draw attention to violations of this Declaration that have not been corrected or that have been made the subject of remedial proceedings or assessments.
- **15.13** Right to Furnish Information. Any Mortgagee can furnish information to the Board concerning the status of any Mortgage.

ARTICLE 16

ARCHITECTURAL CONTROL

16.1 <u>Architectural Restrictions.</u> No building, fence, wall, or other structure shall be commenced, erected or maintained on or within the Project, nor shall any

exterior change, addition or alteration in any structure of the Project, including fences, hedges, patio covers, landscaping, solar or heating systems, spas, pools, awnings, antennas, exterior paint or decor, be commenced, applied, constructed, erected, or maintained by any person other than the Declarant until the plans and specifications showing the nature, kind, shape, height, materials, color, location and approximate cost of the same shall have been submitted to and approved in writing as to the harmony of exterior design, location and relation to surrounding structures and topography by the Architectural Review Board (the "ARB") constituted as set forth in Section 16.2, below. No alteration involving or affecting the structural integrity of any part of the Project shall be made in any Unit or in any portion of the Common Areas surrounding any Unit and no plumbing or electrical work within any bearing or common walls shall be performed by any person until the plans and specifications for such alteration shall have been submitted to and approved in writing by the ARB. If the ARB fails to approve or disapprove such plans and specifications within forty-five (45) calendar days after all of said plans and specifications requested by the ARB have been submitted to it, such approval will not be required, and the Owner proposing such items shall be deemed to have fully complied with this Section. The standards and procedures pursuant to which the ARB shall operate may be set forth by the Ground Lessor in Rules and Regulations issued pursuant to the Common Areas Lease.

- 16.2 Architectural Review Board. There shall be an Architectural Review Board ("ARB") consisting of five (5) members to carry out the functions set forth in this Declaration for that body. The Ground Lessor shall appoint all of the original members of the ARB. Thereafter, the Ground Lessor may, but shall not be obligated to, appoint a majority of the members of the ARB. The Board of Directors shall have the power to appoint any remaining authorized members of the ARB. Any member of the ARB appointed by the Board of Directors shall be a member of the Association. Members appointed by the Ground Lessor need not be members of the Association. All members of the ARB shall be appointed or elected for terms of two (2) years, and those appointed by the Ground Lessor may be removed by the Ground Lessor at any time, for any reason or no reason. An ARB member appointed by the Board of Directors may be removed only for cause. ARB members may be appointed to serve successive terms.
- 16.3 No Waiver. The approval by the ARB of any plans or specifications for any work done or proposed or in connection with any other matter requiring the approval of the ARB or the Board under this Declaration shall not be deemed to be or constitute a waiver of any right to withhold approval as to any similar plan, specification or matter whenever submitted for approval. In addition, the ARB's failure to act shall not be deemed to be or constitute a waiver of any right to review any subsequent building plans without the purview of the ARB.

- 16.4 <u>Disclaimer of Liability.</u> Neither the Board, the Ground Lessor, the ARB, nor any member thereof shall be liable to any Owner for any damage, loss or prejudice suffered or claimed on account of:
- (i) the approval or disapproval of plans or specifications, whether or not defective;
- (ii) the construction or performance of any work, whether or not pursuant to approved plans or specifications; or
- (iii) the development or manner of development of any property within the Project; provided, however, that such member has acted in good faith.
- **16.5** <u>Final Action.</u> Any action taken by the ARB shall be final, subject only to a rehearing which may be granted by the ARB.
- 16.6 <u>Variances.</u> Where circumstances, such as topography, location or property lines, location of trees, or other matters require, the ARB may allow reasonable variances as to any of the covenants, conditions or restrictions contained in this Declaration under the jurisdiction of the ARB on such terms and conditions as it shall require; provided, however, that all such variances shall be in keeping with the general plan for the improvement and development of the Project. Such variances shall not waive the right of the ARB to require strict adherence to this Declaration in all other circumstances.

ARTICLE 17

AMENDMENT OF DECLARATION

- 17.1 <u>Amendment Before Close of First Sale.</u> Before the close of the first sale of a Condominium in the Project to a purchaser other than Declarant, this Declaration and any amendments to it may be amended in any respect or revoked by the execution of an instrument amending or revoking the Declaration by Declarant and any Mortgagee of record of the property described in the Declaration. The amending or revoking instrument shall make appropriate reference to this Declaration and, if this Declaration has been recorded, shall be acknowledged and recorded in the Official Records of the County Recorder of Santa Cruz County, California.
- 17.2 <u>Amendment After Close of First Sale.</u> After the close of the first sale of a Condominium to a purchaser other than Declarant, this Declaration may be amended or revoked in any respect by the vote or written consent of the holders of not less than

seventy-five percent (75%) of all the votes entitled to be cast by each class of Owners, or if a single class of Owners is then in effect, by the vote or written consent of not less than (i) seventy-five percent (75%) of all the votes entitled to be cast by all Owners including Declarant and (ii) fifty-one percent (51%) of the votes entitled to be cast by all Owners excluding Declarant. Consent of the Ground Lessor shall be required prior to the effectiveness of any amendment to this Declaration. Such consent shall not be unreasonably withheld and shall be deemed given if the Ground Lessor shall fail to consent to or reject a proposed amendment for forty-five (45) calendar days after notice thereof has been duly given. Also, if the consent or approval of any governmental authority, Mortgagee, or other person is required under this Declaration with respect to any amendment or revocation of any provision of this Declaration, no such amendment or revocation shall become effective unless such consent or approval is obtained. Any amendment or revocation subsequent to the close of the first sale shall be evidenced by an instrument certified by the secretary or other duly authorized officer of the Association and shall make appropriate reference to this Declaration and its amendments and shall be acknowledged and recorded in the Official Records of the County Recorder of Santa Cruz County, California.

- 17.3 <u>Reliance on Amendments.</u> Any amendments made in accordance with the terms of this Declaration shall be presumed valid by anyone relying on them in good faith.
- 17.4 Amendments to Conform to Secondary Market Requirements. It is the intent of Declarant that the Declaration Instruments, and the Project in general, shall now and in the future meet all requirements necessary to purchase, guarantee, insure, or subsidize any Mortgage of a Declaration in the Project by the Federal Home Loan Mortgage Corporation and/or the Federal National Mortgage Association. The Association and the Owners shall, during the Declarant Control Period, adopt any resolutions required by Declarant to conform this Declaration to the requirements of any of these entities or agencies, subject to the requirements of the Common Areas Lease.

17.5 Prohibited Amendments.

- (a) Notwithstanding any other provisions of this Declaration, the Articles or the Bylaws, no amendment to this Declaration may, without Declarant's consent:
- (i) modify the provisions of Article 10 of this Declaration relating to the amount or types of insurance the Association is required to obtain and maintain;

- (ii) modify the provisions of Article 11 of this Declaration relating to the destruction of improvements and the decision to rebuild or purchase destroyed Units;
- (iii) modify the provisions of Section 9.13(c) relating to the right of Declarant and any Mortgagee to pay delinquent assessments;
- (iv) modify the provisions of Articles 13 or 14 relating to special restrictions on the possession, occupancy, use and resale of Condominiums; or
 - (v) modify this Section 17.5.
- (b) No amendment to this Declaration may violate any provision of the Common Areas Lease.

ARTICLE 18

TERMINATION OF DECLARATION

The terms, covenants, conditions and restrictions set forth in this Declaration shall continue in full force and effect until the expiration of the Common Areas Lease, including all extensions or renewals thereof, or until this Declaration is revoked pursuant to Article 17.

ARTICLE 19

MISCELLANEOUS PROVISIONS

- **19.1** <u>Future Construction.</u> Nothing in this Declaration shall limit the right of Declarant to complete construction of improvements to the Common Areas and to the Condominiums in accordance with the Condominium Plan.
- 19.2 <u>Assignment of Declarant's Rights.</u> The rights of Declarant in this Declaration may be assigned by Declarant to any successor to all or any part of any of Declarant's interests in the Project by an express assignment incorporated in a recorded deed or to a Mortgagee acquiring Declarant's interest in the Project by foreclosure or by deed-in-lieu of foreclosure.
- 19.3 <u>Captions.</u> The captions used in this Declaration are for convenience only and in no way define, limit, or describe the scope of this Declaration or the intent of any provision thereof.

- 19.4 <u>Severability</u>. The provisions of this Declaration shall be deemed independent and severable, and the invalidity or partial invalidity or unenforceability of any provision or provisions shall not invalidate any of the other provisions or part thereof.
- **19.5** <u>Cumulative Remedies; Waiver.</u> Each remedy provided for in this Declaration shall be cumulative and not exclusive. Failure to exercise any remedy provided for in this Declaration shall not, under any circumstances, be construed as a waiver of the remedy.
- 19.6 <u>Violations as Nuisances.</u> Every act or omission in violation of the provisions of this Declaration shall constitute a nuisance and, in addition to all other remedies set forth, may be abated or enjoined by the Association.
 - **19.7** [RESERVED].
- 19.8 <u>Construction.</u> The provisions of this Declaration shall be liberally construed to effectuate its purpose and shall be interpreted under the laws of the State of California. This Declaration and all of the other Condominium Instruments are intended to comply with all of the applicable provisions of the Act, and shall be so interpreted and applied.
- **19.9** <u>Waiver.</u> Failure to enforce any provision of this Declaration shall not constitute a waiver of the right to enforce the provision thereafter.
- 19.10 Notices. All notices, demands, bills, statements, or other communications pursuant to this Declaration shall be in writing and shall be deemed to have been duly given if delivered personally or sent by United States mail (first class, registered or certified), postage prepaid, or if notification is of a default or lien, sent by registered or certified United States mail, return receipt requested, postage prepaid, (i) if to an Owner other than Declarant, at the address which the Owner shall designate in writing and file with the Association or its designated agent or, if no such address is designated, at the address of the Unit of such Owner, or (ii) if to Declarant, at the principal office of Declarant at the Condominium or at such other address as shall be designated by notice in writing to the Owners pursuant to this Section. If a Condominium is owned by more than one person, each such person who so designates an address in writing to the Association shall be entitled to receive all notices hereunder.
- **19.11** Notice of Transfer. Concurrently with the consummation of the sale of any Condominium under circumstances where the transferee becomes an Owner of the Condominium or within five (5) business days thereafter, the transferee shall notify the

Association in writing of such sale. Such notification shall set forth the name of the transferee and his or her Mortgagee and transferor, the address of the Condominium purchased by the transferee, the transferee's and the Mortgagee's mailing address, and the date of sale. Before the receipt of such notification, any and all communications required or permitted to be given by the Association or the Board shall be deemed to be duly made and given to the transferee if duly and timely made and given to the transferee's transferor.

- **19.12** Gender and Number. The use of the masculine gender in this Declaration shall be deemed to include the feminine and neuter genders and the use of the singular shall be deemed to include the plural, and vice versa, whenever the context so requires.
- **19.13** <u>Binding Effect.</u> This Declaration shall inure to the benefit of and be binding on the successors and assigns of Declarant, and the heirs, personal representatives, grantees, tenants, successors, and assigns of all other Owners.
- 19.14 <u>Limitation on Obligations.</u> Nothing contained in this Declaration or elsewhere in the Condominium Instruments shall be deemed to impose upon Declarant or its successors or assigns any obligation of any nature to build, construct, or provide any buildings or improvements. All obligations upon Declarant of this nature shall arise only from a duly executed contract of sale signed by Declarant and a prospective Owner other than Declarant.

Declarant has executed this instrument as of the date first above written.

Ву:			
Its:			

THE REGENTS OF THE UNIVERSITY OF CALIFORNIA, a California public corporation

EXHIBIT A to DECLARATION OF CC&RS

Description of Land

All that certain real property situated in the City of Santa Cruz, County of Santa Cruz, State of California, of First Amended Map Hagar Court (Condominium Plan) A Condominium Subdivision filed on October 17, 2003, in Volume 104 of Maps at page 31 in the Office of the County Recorder, Santa Cruz County, California.

EXHIBIT B to DECLARATION OF CC&RS

Condominium Plan (Not Attached for Recording)

EXHIBIT C to DECLARATION OF CC&RS

Assignment of Percentage Interest

Unit	Square	Interest	Unit	Square	Interest	Unit	Square	Interest
Number	Footage		Number	Footage		Number	Footage	
101	961	.0198	121	961	.0198	141	961	.0198
102	961	.0198	122	874	.0180	142	961	.0198
103	961	.0198	123	874	.0180	143	961	.0198
104	961	.0198	124	961	.0198	144	874	.0180
105	961	.0198	125	961	.0198	145	961	.0198
106	961	.0198	126	961	.0198	146	961	.0198
107	874	.0180	127	961	.0198	147	961	.0198
108	961	.0198	128	961	.0198	148	961	.0198
109	961	.0198	129	961	.0198	149	961	.0198
110	961	.0198	130	874	.0180	150	961	.0198
111	961	.0198	131	961	.0198	Unit 51	917	.0189
112	961	.0198	132	961	.0198	Total	48445	1.000
113	961	.0198	133	961	.0198			
114	961	.0198	134	961	.0198			
115	961	.0198	135	961	.0198			
116	961	.0198	136	961	.0198			
117	961	.0198	137	874	.0180			
118	961	.0198	138	961	.0198			
119	961	.0198	139	961	.0198			
120	961	.0198	140	961	.0198			

****** REFERENCE PAGE – DO NOT INCLUDE IN FINAL DOCUMENT****

EXHIBITS

Exhibit A: Description of Land Exhibit B: Condominium Plan

Exhibit C: Assignment of Percentage Interest

COVENANTS, CONDITIONS & RESTRICTIONS

FOR HAGAR COURT CONDOMINIUMS

CITY OF SANTA CRUZ, COUNTY OF SANTA CRUZ STATE OF CALIFORNIA

		rage
DECLARAT	TION OF COVENANTS, CONDITIONS & RESTRICTIONS	1
ARTICLE 1	DEFINITIONS, EXHIBITS AND INTERPRETATION	2
1.1	Definitions. As used in this Declaration:	2
1.2	Exhibits	6
1.3	Interpretation of Declaration	6
ARTICLE 2	ESTABLISHMENT OF THE CONDOMINIUM	7
2.1	Submission of Property and Division into Estates	7
2.2	Recordation of Condominium Plan	8
2.3	Assignment of Percentage Interest	8
ARTICLE 3	UNITS, EASEMENTS AND RESERVATIONS	8
3.1	Units	8
3.2	Nonexclusive Easements	9
3.3	Exclusive Use Common Areas	11
3.4	Encroachments	11
3.5	Power to Grant Easements	11
3.6	Reservations for Sales and Management Purposes	12
3.7	Reservation for Rental Purposes	13
ARTICLE 4	USE RESTRICTIONS	13
4.1	Business Usage Prohibited	13
4.2	Liquid Furniture	14
4.3	[NOT USED]	14
4.4	Signs and Flags	14
4.5	Children	14
4.6	Pets	14
4.7	Offensive Activities	14
4.8	Owner Negligence	15
4.9	Trash Disposal	15
4.10	Outside Laundering and Drying	15
4.11	Prohibited Restrictions	15
4.12	Storage	15

(continued)

Page

4.13	Antenna and Clothesline	15
4.14	Personal Business Records Permitted	16
4.15	Vehicles and Parking	16
4.16	Restrictions Applicable to Guests, etc	16
4.17	Fences and Screens	16
4.18	Gas or Liquid Storage	16
4.19	Potted Plants	16
4.20	Diseased Plants	16
4.21	Climbing Vines	17
4.22	Plants in Common Areas	17
4.23	Alterations	17
4.24	Effect on Insurance	17
4.25	Delegation of Use	17
4.26	Single Family Use	18
4.27	Indemnification	18
4.28	Enforcement and Remedies	18
ARTICLE 5	MAINTENANCE AND ALTERATIONS	18
5.1	Owner's Responsibilities	18
5.2	Association's Responsibilities	19
ARTICLE 6	TAXES AND ASSESSMENTS	19
6.1	Separate Assessments	19
6.2	Payment Prior to Segregation	20
ARTICLE 7	THE ASSOCIATION	20
7.1	Incorporation	20
7.2	Action By Board, Officers, and Designees	20
7.3	General Statement of Powers	20
7.4	Rights of Assessment	21
7.5	Right To Establish Rules	21
7.6	Right To Impose Sanctions	21
7.7	Right of Delegation	22

(continued)

		Page
7.8	General Statement of Duties	22
7.9	Repair and Maintenance of Common Areas and Exclusive Use Common	22
7.10	Areas Contracts for Goods and Services	
7.10	Payment of Taxes and Assessments	
7.11	Furnishing of Utilities	
7.12	Securing Insurance	
7.14	Preparation and Distribution of Financial Information and Governing Instruments	
7.15	Other Duties	25
7.16	Limitations on Authority of Board and the Association	25
7.17	Limitation on Liability of Directors and Officers	26
7.18	Inspection of Books and Records	26
ARTICLE 8	ASSOCIATION MEMBERSHIP AND VOTING	27
8.1	Membership Appurtenant to Ownership	27
8.2	Approval by a Specified Percentage of Owners	28
8.3	Voting	28
8.4	Joint Ownership Votes	29
8.5	Cumulative Voting	29
ARTICLE 9	ASSESSMENTS	30
9.1	Payment of Assessments	30
9.2	Initial Working Capital Payment	30
9.3	Limited Exemption During Construction	31
9.4	Personal Obligation of Owner	31
9.5	Scope of Assessment Authority	31
9.6	Regular Assessments	32
9.7	Special Assessments	32
9.8	Allocation of Assessments	33
9.9	Fiscal Year and Assessment Period	33
9.10	Due Date, Late Charges, and Interest	34
9.11	Estoppel Certificates	34

(continued)

Page

9.12	Enforcement of Assessments	35
9.13	Creation of Assessment Lien	35
9.14	Foreclosure Under Assessment Lien	36
9.15	Waiver of Homestead Protection	37
9.16	Common Areas Interest	37
ARTICLE 1	0 INSURANCE	37
10.1	Liability Insurance	37
10.2	Fire and Casualty and Demolition Insurance	38
10.3	Appointment of a Trustee	38
10.4	Owner's Insurance	39
10.5	Workers' Compensation and Other Association Insurance	39
10.6	Adjustment of Losses	40
10.7	Distribution to Mortgagees	40
10.8	Director and Officer Liability Insurance	40
10.9	Fidelity	41
10.1	0 Limitation of Liability for Failure to Obtain Insurance	41
10.1	1 Notice to Owners	41
10.1	2 Generally Applicable Requirements	41
10.1	3 Umbrella Liability Insurance	42
10.1	4 Increases in Coverage	42
ARTICLE 1	1 DESTRUCTION OF IMPROVEMENTS	42
11.1	Minor Repair and Reconstruction	42
11.2	Destruction When Insurance Proceeds Exceed Eighty-Five Percent (85%) of Reconstruction Costs	42
11.3	Destruction When Insurance Proceeds Are Less Than Eighty-Five Percent (85%) of Reconstruction Costs	43
11.4	Repair and Reconstruction of Improvements	43
11.5	Purchase After Election Not To Rebuild; Restoration of Site	44
11.6	Application of Insurance Proceeds	45
11.7	Termination of Partition Waiver	45
11.8	Distribution of Excess Insurance Proceeds	46

(continued)

		Page
ARTICLE 12	CONDEMNATION; PARTITION	46
12.1	Condemnation	47
12.2	Waiver of Right to Partition	47
ARTICLE 13	SPECIAL RESTRICTIONS UPON POSSESSION, OCCUPANCY AND USE OF UNITS	47
13.1	Purpose	47
13.2	[RESERVED]	47
13.3	[RESERVED]	47
13.4	Qualified Use	47
13.5	Qualified Person	47
13.6	Conditionally Qualified Persons	48
13.7	Offer or Sale by Conditionally Qualified Person	49
13.8	Qualified Users	49
13.9	Enforcement of Qualified Use Restriction	50
ARTICLE 14	LIMITATIONS ON TRANSFER OF CONDOMINIUMS	50
14.1	Right of First Refusal	50
14.2	Purchase Price and Terms	51
14.3	Closing Payment	53
14.4	Secured Parties	53
14.5	Approved Purchasers	54
14.6	Certification of Status	54
14.7	Leasing	54
14.8	Unauthorized Sales or Transfers	55
14.9	Termination of Articles 13 and 14	55
14.10	Enforcement of Article 14	55
14.11	Effect of Article 14	55
ARTICLE 15	PROTECTION OF MORTGAGEES	56
15.1	Mortgage Permitted	56
15.2	Notice to Board of Directors	56

(continued)

Page

	15.3	Subordination	56
	15.4	Control of Amendment of Condominium Instruments	56
	15.5	Restrictions on Certain Changes	58
	15.6	Right to Examine Books and Records	59
	15.7	Priority in Distribution of Proceeds	59
	15.8	Status of Improvements	59
	15.9	Notices to Mortgagees	59
	15.10	Lien Not Invalidated	60
	15.11	Status of Loan to Facilitate Resale	60
	15.12	Right to Appear at Meetings	60
	15.13	Right to Furnish Information	60
ARTIC	LE 16	ARCHITECTURAL CONTROL	60
	16.1	Architectural Restrictions	60
	16.2	Architectural Review Board	61
	16.3	No Waiver	61
	16.4	Disclaimer of Liability	62
	16.5	Final Action	62
	16.6	Variances	62
ARTIC	LE 17	AMENDMENT OF DECLARATION	62
	17.1	Amendment Before Close of First Sale	62
	17.2	Amendment After Close of First Sale	62
	17.3	Reliance on Amendments	63
	17.4	Amendments to Conform to Secondary Market Requirements	63
	17.5	Prohibited Amendments	63
ARTIC	LE 18	TERMINATION OF DECLARATION	64
ARTIC	LE 19	MISCELLANEOUS PROVISIONS	65
	19.1	Future Construction	65
	19.2	Assignment of Declarant's Rights	65
	19.3	Captions	64
	19.4	Severability	65

(continued)

		Page
19.5	Cumulative Remedies; Waiver	65
19.6	Violations as Nuisances	65
19.7	[RESERVED]	65
19.8	Construction	65
19.9	Waiver	65
19.10	Notices	65
19.11	Notice of Transfer	65
19.12	Gender and Number	66
19.13	Binding Effect	66
19.14	Limitation on Obligations	66

EXHIBITS

Exhibit A: Description of Land Exhibit B: Condominium Plan

Exhibit C: Assignment of Percentage Interest